UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): August 4, 2023

7GC & Co. Holdings Inc. (Exact name of registrant as specified in its charter)

Delaware	001-39826	85-3118980
(State or other jurisdiction of incorporation)	(Commission File Number)	(I.R.S. Employer Identification No.)
388 Market Street, Suite 1300 San Francisco, CA		94111
(Address of Principal Executive Office	ces)	(Zip Code)
(Regi	(628) 400-9284 strant's telephone number, including area o	code)
Check the appropriate box below if the Form 8-K is in provisions:	ntended to simultaneously satisfy the filing o	obligation of the registrant under any of the following
☑ Written communication pursuant to Rule 425 under	the Securities Act (17 CFR 230.425)	
☐ Soliciting material pursuant to Rule 14a-12 under the	he Exchange Act (17 CFR 240.14a-12)	
☐ Pre-commencement communications pursuant to R	ule 14d-2(b) under the Exchange Act (17 CFF	R 240.14d-2(b))
☐ Pre-commencements communications pursuant to F	Rule 13e-4(c) under the Exchange Act (17 CF	R 240.13e-4(c))
Securities registered pursuant to Section 12(b) of the Ac	rt:	
Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Units, each consisting of one share of Class A Common Stock, par value \$0.0001 per share, and one-half Redeemable Warrant	VIIAU	The Nasdaq Stock Market LLC
Shares of Class A common stock, par value \$0.0001 per share	VII	The Nasdaq Stock Market LLC
Redeemable Warrants, each whole warrant exercisable for one share of Class A Common Stock at an exercise price of \$11.50	VIIAW	The Nasdaq Stock Market LLC
Indicate by check mark whether the registrant is an enchapter) or Rule 12b-2 of the Securities Exchange Act of		e 405 of the Securities Act of 1933 (§230.405 of this
Emerging growth company \boxtimes		
If an emerging growth company, indicate by check mar or revised financial accounting standards provided pursu		extended transition period for complying with any new

Item 1.01. Entry Into a Material Agreement.

On August 4, 2023, 7GC & Co. Holdings Inc. ("<u>7GC</u>") and Banzai International, Inc. ("<u>Banzai</u>") entered into an amendment (the "<u>Amendment</u>") to their previously disclosed Agreement and Plan of Merger and Reorganization, dated as of December 8, 2022 (as amended, the "<u>Merger Agreement</u>"), among 7GC, Banzai and the other parties thereto, pursuant to which 7GC and Banzai agreed to enter into a business combination transaction (the "<u>Business Combination</u>"). Upon consummation of the Business Combination, Banzai will become a wholly owned subsidiary of 7GC, 7GC will change its name to Banzai International, Inc. and the securities of the combined company are expected to be listed on the Nasdaq Capital Market.

As previously disclosed in the Current Report on Form 8-K filed by 7GC with the Securities and Exchange Commission (the "SEC") on December 8, 2022, prior to the execution and delivery of the Merger Agreement, on December 8, 2022, Banzai entered into that certain Agreement and Plan of Merger (the "Hyros Purchase Agreement") with a direct, wholly owned subsidiary of Banzai (the "Hyros Merger Sub"), Hyros Inc. ("Hyros") and the stockholder representative party thereto, pursuant to which Hyros Merger Sub would have merged with and into Hyros, and Hyros would have continued as the surviving company and as a wholly owned subsidiary of Banzai (the "Hyros Acquisition"). On August 1, 2023, Banzai and Hyros agreed to terminate the Hyros Purchase Agreement. No termination fee was paid or is payable in connection with such termination of the Hyros Purchase Agreement.

Pursuant to the Amendment, 7GC and Banzai agreed to amend the terms and conditions of the Merger Agreement as follows:

- the closing (the "Closing") of the transactions contemplated by the Merger Agreement (the "Transactions") is no longer conditioned upon the consummation of the Hyros Acquisition;
- the value of the total consideration payable to Banzai stockholders (which will consist solely of newly issued shares of 7GC common stock) is reduced from \$293.0 million to \$100.0 million, with no post-closing "earn-out" or other future contingent consideration;
- the "Termination Date" upon which either party may terminate the Merger Agreement for any reason (subject to certain conditions set forth in the Merger Agreement) if the closing of the Transactions has not yet occurred is extended from September 8, 2023 to December 28, 2023;
- the \$100.0 million minimum aggregate transaction proceeds condition is replaced with a \$5.0 million minimum net cash condition, pursuant to which Banzai's obligation to consummate the Transactions is subject to a condition that the "Net Transaction Proceeds" (as defined below) equal or exceed \$5.0 million, with "Net Transaction Proceeds" being defined as being an amount equal to (i) the sum of (A) the cash proceeds to be received by 7GC at Closing from the trust account established in connection with 7GC's initial public offering ("IPO") (after giving effect to redemptions by 7GC stockholders), (B) the cash proceeds to be received by 7GC or any of Banzai or its subsidiaries from any financing, whether equity of debt, at or immediately following the Closing, and (C) the unrestricted cash on the balance sheet of Banzai as of immediately prior to the Closing, minus (ii) the 7GC Transaction Expenses (as defined in the Amendment), minus (iii) the Company Expenses (as defined in the Amendment);
- the conversion formula for the conversion of Banzai SAFE Rights (as defined below) into shares of 7GC Class A common stock upon Closing is revised to provide that each right (each, a "SAFE Right") pursuant to certain existing Banzai Simple Agreements for Future Equity (each, a "SAFE Agreement") that is outstanding immediately prior to the Closing will be cancelled and converted into the right to receive a number of newly issued shares of 7GC Class A common stock equal to (i) the SAFE Purchase Amount (as defined in the applicable SAFE Agreement that governs such SAFE Right) in respect of such SAFE Right, *divided by* the SAFE Conversion Price (the Valuation Cap Price as defined in the applicable SAFE Agreement that governs such SAFE Right) in respect of such SAFE Right, *multiplied by* (ii) the Exchange Ratio;
- the deadline for Banzai to deliver its PCAOB audited financial statements to 7GC for inclusion in 7GC's Registration Statement on Form S-4 in connection with the Transactions is extended to August 31, 2023; and
- the form of Amended and Restated Registration Rights Agreement to be entered into at Closing by 7GC, 7GC & Co. Holdings LLC (the "Sponsor") and certain stockholders of the Company is amended and restated in its entirety, to be in the form attached as Exhibit B to the Amendment.

The form of Amended and Restated Registration Rights Agreement was amended and restated to amend the lock-up period that will be applicable to the Sponsor and certain other parties holding the 5,750,000 shares of 7GC's Class B common stock issued in connection with 7GC's initial public offering from (x) the earlier of (A) one year after the date of Closing, (B) the first date the closing price of 7GC's common stock equals or exceeds \$12.00 per share for any 20 trading days within any 30 trading day period commencing at least 150 days after the date of Closing, and (C) the date on which 7GC completes a liquidation, merger, or other reorganization or similar transaction that results in all of 7GC's stockholders having the right to exchange their shares of 7GC common stock for cash, securities or other property to (y) the earlier of (A) 180 days after the Closing or (B) the date on which 7GC completes a liquidation, merger, or other reorganization or similar transaction that results in all of 7GC's stockholders having the right to exchange their shares of 7GC common stock for cash, securities or other property. The foregoing description of the Amended and Restated Registration Rights Agreement is not complete and is qualified in its entirety by reference to the form of Amended and Restated Registration Rights Agreement, a form of which is filed as Exhibit 10.2 to this Current Report on Form 8-K (this "Current Report") and incorporated herein by reference.

The foregoing description of the Amendment is not complete and is qualified in its entirety by reference to the Amendment, which is filed as Exhibit 2.1 to this Current Report and incorporated herein by reference.

Concurrently with the execution and delivery of the Amendment, on August 4, 2023, 7GC entered into an Amended and Restated Company Support Agreement (the "Amended and Restated Company Support Agreement") with Banzai and certain stockholders of Banzai set forth on Schedule I thereto (the "Company Stockholders"), pursuant to which the Company Stockholders reiterated their agreement not to (i) sell, assign, offer, exchange, transfer, pledge, dispose of, permit to exist any lien, security interest, or similar encumbrance with respect to, or otherwise encumber, any of the shares held by the Company Stockholders set forth on Schedule I thereto (the "Subject Shares"), (ii) deposit any Subject Shares into a voting trust or enter into a voting agreement or arrangement or grant any proxy or power of attorney with respect thereto that is inconsistent with the Amended and Restated Company Support Agreement, (iii) enter into any contract, option or other arrangement or undertaking with respect to the direct or indirect acquisition or sale, assignment, transfer or other disposition of the Subject Shares, or (iv) take any action that would have the effect of preventing, impeding, interfering or adversely affecting such Company Stockholder's ability to perform its obligations under the Amended and Restated Company Support Agreement. Additionally, each Company Stockholder, during the period commencing on the date of the Amended and Restated Company Support Agreement until the Closing or termination of the Merger Agreement, reiterated its agreement that, at any meeting of the stockholders of Banzai, and in any action by written consent of the stockholders of Banzai, such Company Stockholder will, if a meeting is held, appear at the meeting and vote or provide consent, in person or by proxy, to vote all of its, his or her Subject Shares: (a) to approve and adopt the Merger Agreement and the Transactions; (b) in any other circumstances upon which a consent or other approval is required under the governing documents of Banzai or Banzai stockholder agreements or otherwise sought with respect to, or in connection with, the Merger Agreement or the Transactions, to vote, consent or approve (or cause to be voted, consented or approved) with respect to all of such Company Stockholder's Subject Shares held at such time in favor thereof; and (c) to vote, consent or disapprove (or cause to be voted, consented or disapproved) with respect to all of such Company Stockholder's Subject Shares held at such time against any 7GC Competing Transaction (as defined in the Merger Agreement) or any proposal, action or agreement that would impede, interfere, frustrate, delay, postpone, prevent or nullify any provision of the Amended and Restated Company Support Agreement, the Merger Agreement or the Transactions.

The foregoing description of the Amended and Restated Company Support Agreement is not complete and is qualified in its entirety by reference to the Amended and Restated Company Support Agreement, which is filed as Exhibit 10.1 to this Current Report and incorporated herein by reference.

Concurrently with the execution and delivery of the Amendment, on August 4, 2023, the Sponsor entered into a Sponsor Forfeiture Agreement (the "Sponsor Forfeiture Agreement") with 7GC and Banzai, pursuant to which, contingent upon Closing, the Sponsor agreed to forfeit all 7,350,000 of its private placement warrants to purchase shares of 7GC's Class A common stock, exercisable at \$11.50 per share (the "Forfeited Private Placement Warrants"), acquired by the Sponsor in December 2020 in connection with the 7GC initial public offering. At the Closing, the Forfeited Private Placement Warrants shall be transferred from the Sponsor to 7GC for cancellation in exchange for no consideration and 7GC shall retire and cancel all of the Forfeited Private Placement Warrants.

The foregoing description of the Sponsor Forfeiture Agreement is not complete and is qualified in its entirety by reference to the Sponsor Forfeiture Agreement, which is filed as Exhibit 10.3 to this Current Report and incorporated herein by reference.

Item 7.01. Regulation FD Disclosure.

On August 7, 2023, 7GC and Banzai issued a joint press release announcing the execution of the Amendment and the Sponsor Forfeiture Agreement, which is filed as Exhibit 99.1 to this Current Report and incorporated herein by reference.

The information in this Item 7.01, including Exhibit 99.1, is furnished and shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934 (the "Exchange Act"), or otherwise subject to liabilities under that section, and shall not be deemed to be incorporated by reference into the filings of 7GC under the Securities Act of 1933, as amended, or the Exchange Act, regardless of any general incorporation language in such filings.

Item 9.01. Financial Statement and Exhibits.

(d) Exhibits.

and Banzai
nc., Banzai
and Banzai

No Offer or Solicitation

This Current Report does not constitute an offer to sell, or a solicitation of an offer to buy, or a recommendation to purchase, any securities in any jurisdiction, or the solicitation of any vote, consent or approval in any jurisdiction in connection with the Business Combination or any related transactions, nor shall there be any sale, issuance or transfer of any securities in any jurisdiction where, or to any person to whom, such offer, solicitation or sale may be unlawful under the laws of such jurisdiction. This Current Report does not constitute either advice or a recommendation regarding any securities. No offering of securities shall be made except by means of a prospectus meeting the requirements of the Securities Act or an exemption therefrom.

Forward-Looking Statements

Certain statements included in this Current Report are not historical facts but are forward-looking statements, including for purposes of the safe harbor provisions under the United States Private Securities Litigation Reform Act of 1995. Forward-looking statements generally are accompanied by words such as "believe," "may," "will," "estimate," "continue," "anticipate," "intend," "expect," "should," "would," "plan," "project," "forecast," "predict," "potential," "seem," "seek," "future," "outlook," "target," and similar expressions that predict or indicate future events or trends or that are not statements of historical matters, but the absence of these words does not mean that a statement is not forward-looking. These forward-looking statements include, but are not limited to, (1) references with respect to the anticipated benefits of the Business Combination and anticipated closing timing; (2) the sources and uses of funds for the Business Combination, (3) the anticipated capitalization and enterprise value of the combined company following the consummation of the Business Combination; and (4) current and future potential commercial and customer relationships. These statements are based on various assumptions, whether or not identified in this Current Report, and on the current expectations of 7GC's and Banzai's management and are not predictions of actual performance. These forward-looking statements are provided for illustrative purposes only and are not intended to serve as, and must not be relied on by any investor as, a guarantee, an assurance, a prediction or a definitive statement of fact or probability. Actual events and circumstances are difficult or impossible to predict and will differ from assumptions. Many actual events and circumstances are beyond the control of 7GC and Banzai. These forward-looking statements are subject to a number of risks and uncertainties, including: the inability of the parties to timely or successfully complete the Business Combination, including due to redemptions by 7GC's public stockholders, the failure to satisfy the minimum net cash closing condition, the failure to maintain the Nasdaq listing of 7GC's securities, the failure to obtain certain regulatory approvals or the satisfaction of other conditions to closing in the Merger Agreement; changes in domestic and foreign business, market, financial, political and legal conditions; the inability of the parties to successfully or timely consummate the Business Combination, including the risk that any required stockholder or regulatory approvals are not obtained, are delayed or are subject to unanticipated conditions that could adversely affect the combined company; failure to realize the anticipated benefits of the Business Combination; risks relating to the uncertainty of the projected financial information with respect to Banzai; Banzai's ability to successfully and timely develop, sell and expand its technology and products, and otherwise implement its growth strategy; risks relating to Banzai's operations and business, including information technology and cybersecurity risks, loss of customers and deterioration in relationships between Banzai and its employees; risks related to increased competition; risks relating to potential disruption of current plans, operations and infrastructure of Banzai as a result of the announcement and consummation of the Business Combination; risks that the post-combination company experiences difficulties managing its growth and expanding operations; the amount of redemption requests made by 7GC's stockholders; the impact of geopolitical, macroeconomic and market conditions, including the COVID-19 pandemic; the ability to successfully select, execute or integrate future acquisitions into the business, which could result in material adverse effects to operations and financial conditions; and those factors discussed in the sections entitled "Risk Factors" and "Special Note Regarding Forward-Looking Statements" in 7GC's Quarterly Report on Form 10-Q for the quarter ended March 31, 2023, 7GC's Annual Report on Form 10-K for the year ended December 31, 2022, and in those documents that 7GC has filed, or will file, with the SEC. If any of these risks materialize or our assumptions prove incorrect, actual results could differ materially from the results implied by these forward-looking statements. The risks and uncertainties above are not exhaustive, and there may be additional risks that neither 7GC nor Banzai presently know or that 7GC and Banzai currently believe are immaterial that could also cause actual results to differ from those contained in the forward-looking statements. In addition, forward-looking statements reflect 7GC's and Banzai's expectations, plans or forecasts of future events and views as of the date of this Current Report. 7GC and Banzai anticipate that subsequent events and developments will cause 7GC's and Banzai's assessments to change. However, while 7GC and Banzai may elect to update these forward-looking statements at some point in the future, 7GC and Banzai specifically disclaim any obligation to do so. These forward-looking statements should not be relied upon as representing 7GC's and Banzai's assessments as of any date subsequent to the date of this Current Report. Accordingly, undue reliance should not be placed upon the forward-looking statements.

Important Information for Investors and Stockholders

The Business Combination will be submitted to stockholders of 7GC for their consideration and approval at a special meeting of stockholders. 7GC and Banzai will prepare a registration statement on Form S-4 (the "Registration Statement") to be filed with the SEC by 7GC, which will include preliminary and definitive proxy statements to be distributed to 7GC's stockholders in connection with 7GC's solicitation for proxies for the vote by 7GC's stockholders in connection with the Business Combination and other matters as described in the Registration Statement, as well as the prospectus relating to the offer of the securities to be issued to 7GC's stockholders and certain of Banzai's equityholders in connection with the completion of the Business Combination. After the Registration Statement has been filed and declared effective, 7GC will mail a definitive proxy statement and other relevant documents to its stockholders as of the record date established for voting on the Business Combination. 7GC's stockholders and other interested persons are advised to read, once available, the preliminary proxy statement/prospectus and any amendments thereto and, once available, the definitive proxy statement/prospectus, in connection with 7GC's solicitation of proxies for its special meeting of stockholders to be held to approve, among other things, the Business Combination, because these documents will contain important information about 7GC, Banzai and the Business Combination. Stockholders may also obtain a copy of the preliminary or definitive proxy statement, once available, as well as other documents filed with the SEC regarding the Business Combination and other documents filed with the SEC by 7GC, without charge, at the SEC's website located at www.sec.gov. Copies of these filings may be obtained free of charge on 7GC's "SEC Filings" website at www.7gc.holdings/sec-filings or by directing a request to info@7gc.co.

Participants in the Solicitation

7GC and Banzai and their respective directors and executive officers, under SEC rules, may be deemed to be participants in the solicitation of proxies of 7GC's stockholders in connection with the Business Combination. Investors and security holders may obtain more detailed information regarding 7GC's directors and executive officers in 7GC's filings with the SEC, including 7GC's Annual Report on Form 10-K filed with the SEC on March 31, 2023. Information regarding the persons who may, under SEC rules, be deemed participants in the solicitation of proxies to 7GC's stockholders in connection with the Business Combination, including a description of their direct and indirect interests, which may, in some cases, be different than those of 7GC's stockholders generally, will be set forth in the Registration Statement. Stockholders, potential investors, and other interested persons should read the Registration Statement carefully when it becomes available before making any voting or investment decisions.

This Current Report is not a substitute for the Registration Statement or for any other document that 7GC may file with the SEC in connection with the potential Business Combination. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THE DOCUMENTS FILED WITH THE SEC CAREFULLY AND IN THEIR ENTIRETY WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain free copies of other documents filed with the SEC by 7GC through the website maintained by the SEC at www.sec.gov.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

7GC & Co. Holdings Inc.

Date: August 7, 2023 By: /s/ Jack Leeney

Name: Jack Leeney

Title: Chairman and Chief Executive Officer

AMENDMENT TO AGREEMENT AND PLAN OF MERGER AND REORGANIZATION

This Amendment to the Agreement and Plan of Merger and Reorganization, dated as of December 8, 2022 (the "Merger Agreement"), by and among Banzai International, Inc., a Delaware corporation (the "Company"), 7GC & Co. Holdings Inc., a Delaware corporation ("7GC"), 7GC Merger Sub I, Inc., a Delaware corporation and an indirect wholly owned subsidiary of 7GC ("First Merger Sub"), and 7GC Merger Sub II, LLC, a Delaware limited liability company and a direct wholly owned subsidiary of 7GC ("Second Merger Sub" and, together with First Merger Sub, the "Merger Subs" and each, a "Merger Sub"), is made and entered into as of August 4, 2023, by and between the Company and 7GC (this "Amendment"). 7GC and the Company will each be referenced to herein from time to time as a "Party" and, collectively, as the "Parties."

RECITALS

WHEREAS, the Parties have entered into the Merger Agreement, which provides for, among other things, the merger of First Merger Sub with and into the Company, with the Company continuing as the surviving entity (the "<u>First Merger</u>") and promptly following the First Merger, but in any event on the same day as the First Merger and as part of the same overall transaction as the First Merger, the Company, as the surviving entity from the First Merger, will merge with and into Second Merger Sub (the "<u>Second Merger</u>" and, together with the First Merger, the "<u>Mergers</u>"), with the separate corporate existence of the Company ceasing and Second Merger Sub surviving the Second Merger as a direct wholly owned subsidiary of 7GC;

WHEREAS, pursuant to Section 8.3 of the Merger Agreement, the Merger Agreement may be amended or modified only by a written agreement executed and delivered by duly authorized officers of 7GC, on the one hand, and the Company, on the other hand; and

WHEREAS, the Parties desire to amend the Merger Agreement as set forth in this Amendment.

NOW THEREFORE, in consideration of the foregoing and the mutual covenants contained in this Amendment, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

AGREEMENT

- 1. Amendment. From and after the date hereof, the Merger Agreement shall be amended as follows:
 - 1.1 The fourth recital of the Merger Agreement is hereby amended and restated in its entirety to read as follows:

"WHEREAS, the Company has entered into that certain Agreement and Plan of Merger, dated as of December 8, 2022 (as may be amended from time to time, the "Hyros Merger Agreement"), by and among the Company, Hero Merger Sub, Inc., a Delaware corporation and a direct wholly owned subsidiary of the Company, Hyros Inc., a Delaware corporation ("Hyros"), and the stockholder representative party thereto;"

- 1.2 The sixth recital of the Merger Agreement is hereby amended by deleting the following language before clause (i) therein in its entirety:
 - "subject to the terms and conditions of this Agreement, immediately following the consummation of the Hyros Acquisition, and"
- 1.3 <u>Section 1.1</u> of the Merger Agreement is hereby amended by:
 - (a) amending and restating the definition of "7GC Transaction Expenses" in its entirety to read as follows:
- ""7GC Transaction Expenses" means the aggregate amount of unpaid out-of-pocket fees, commissions, costs and expenses due and payable by 7GC as of the Second Effective Time of outside legal counsel, accountants, advisors, investment bankers or consultants in connection with the negotiation, preparation, execution and performance of this Agreement or any Ancillary Document and the consummation of the transactions contemplated hereby and thereby (including 7GC's IPO)."
 - (b) deleting the definition of "Aggregate SAFE Cash-Out Amount" in its entirety.
 - (c) replacing the definition of "Aggregate 7GC Transaction Proceeds" with the following in its entirety:
- "Aggregate Transaction Proceeds" means an amount equal to the sum of (a) the cash proceeds to be received by 7GC at Closing from the Trust Account in connection with the transactions contemplated hereby (after, for the avoidance of doubt, giving effect to the 7GC Stockholder Redemptions), (b) the cash proceeds to be received by 7GC or any Group Company from any financing, whether equity or debt, at or immediately following the Closing and (c) the unrestricted cash on the balance sheet of the Company as of immediately prior to the Closing."
 - (d) amending and restating the definition of "Closing Adjusted Vested Securities Merger Consideration" in its entirety to read as follows:
 - ""Closing Adjusted Vested Securities Merger Consideration" means an amount equal to the Total Consideration."
 - (e) deleting the definition of "Closing Vested Securities Merger Consideration" in its entirety.
 - (f) amending and restating the definition of "Company Expenses" in its entirety to read as follows:

""Company Expenses" means, without duplication, the aggregate unpaid amount due and payable as of the Second Effective Time by any Group Company, for (i) out-of-pocket fees, costs and expenses incurred in connection with the negotiation, preparation or execution of this Agreement or any Ancillary Documents and the consummation of the transactions contemplated hereby and thereby (including the fees and expenses of outside legal counsel, accountants, advisors, investment bankers, brokers, consultants or other agents (including, for the avoidance of doubt, to perform any compensation studies)), (ii) the cost of the Company D&O Tail Policy to be obtained pursuant to Section 5.5, (iii) the costs and expenses of any consultant or advisor engaged to prepare a compensation study in connection with implementation of the New Incentive Plan, (iv) any filing fee to be paid pursuant to the HSR Act, (v) the filing fee to be paid for the Registration Statement / Proxy Statement, and (vi) any other fees, expenses, commissions or other amounts that are expressly allocated to any Group Company, or the Pre-Closing Holders pursuant to this Agreement or any Ancillary Document, in each case as of the Second Effective Time."

- (g) amending and restating the definition of "Excess 7GC Transaction Expenses" in its entirety to read as follows:
- ""Excess 7GC Transaction Expenses" means \$0.00."
- (h) deleting the definition of "Hyros Acquisition";
- (i) adding the following definition of "Net Transaction Proceeds":
- ""Net Transaction Proceeds" means an amount equal to (a) the Aggregate Transaction Proceeds, minus (b) the 7GC Transaction Expenses, minus (c) the Company Expenses."
 - (j) amending the definition of "Subsidiary" to remove the last sentence therein in its entirety;
 - (k) adding the following definition of SAFE Conversion Price:
 - ""SAFE Conversion Price" with respect to each SAFE Right, means the Valuation Cap Price as defined in each SAFE Agreement."
 - (l) adding the following definition of "SAFE Purchase Amount":
- "SAFE Purchase Amount" means with respect to each SAFE Right, the Purchase Amount as defined in the applicable SAFE Agreement that governs such SAFE Right."
 - (m) amending and restating the definition of "Total Company Vested Securities" in its entirety to read as follows:
- ""Total Company Vested Securities" means (i) the total number of shares of Company Class A Common Stock and shares of Company Class B Common Stock issued and outstanding as of immediately prior to the First Effective Time (including, for the avoidance of doubt, shares of Company Class A Common Stock issued in connection with the Company Preferred Conversion), (ii) the maximum aggregate number of shares of Company Class A Common Stock issuable upon full exercise of all Company Options issued, outstanding and vested immediately prior to the First Effective Time, (iii) the maximum aggregate number of shares of Company Class A Common Stock issuable upon conversion of the Conversion Price, (iv) the maximum aggregate number of shares of Company Class A Common Stock issuable upon conversion of the Outstanding Amount under each Subordinated Convertible Note as of immediately prior to the First Effective Time at the applicable Subordinated Convertible Note Conversion Price and (v) the maximum aggregate number of shares of Company Class A Common Stock issuable upon conversion of the SAFE Purchase Amount under each SAFE Right as of immediately prior to the First Effective Time at the applicable SAFE Conversion Price."

- (n) amending and restating the definition of "Total Consideration" in its entirety to read as follows:
- ""Total Consideration" means \$100,000,000, payable in 7GC New Class A Shares or 7GC New Class B Shares, as applicable."
- 1.4 The first sentence of Section 2.1(a) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:
- "Upon the terms and subject to the conditions set forth in this Agreement, 7GC, First Merger Sub, Second Merger Sub and the Company (First Merger Sub and the Company sometimes being referred to herein as the "First Merger Constituent Corporations") shall cause First Merger Sub and the Company to consummate the First Merger, pursuant to which the First Merger Sub shall be merged with and into the Company, with the Company being the surviving corporation."
- 1.5 <u>Section 2.2(a)(i)</u> of the Merger Agreement is hereby amended by replacing the language in the first parentheticals therein with the following in its entirety:
 - "including, for the avoidance of doubt, each share of Restricted Stock and each share of Company Class A Common Stock converted from shares of Company Preferred Stock in the Company Preferred Conversion"
 - 1.6 Section 2.2(c) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:
 - "Treatment of SAFE Rights. At the First Effective Time, by virtue of the First Merger and without any action on the part of any Party or any other Person, each SAFE Right that is outstanding immediately prior to the First Effective Time shall thereupon by cancelled and converted into and become (A) the right to receive a number of 7GC New Class A Shares equal to (1) the SAFE Purchase Amount in respect of such SAFE Right *divided by* the SAFE Conversion Price in respect of such SAFE Right *multiplied by* (2) the Exchange Ratio, *plus* (B) the right to receive the Earn Out Shares."
 - 1.7 <u>Section 2.2(e)</u> of the Merger Agreement is hereby amended and restated in its entirety to read as follows:
 - "Aggregate Merger Consideration. The aggregate consideration to be paid to the Pre-Closing Holders in the First Merger shall consist of the Closing Merger Consideration apportioned pursuant to Section 2.2(a), Section 2.2(b) and Section 2.2(c) and Section 2.1(d)."

- 1.8 The first paragraph of Section 2.6(a) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:
- "(a) At the First Effective Time, in accordance with the provisions of <u>Section 2.2(b)</u> and the Allocation Schedule, 7GC shall issue or cause to be issued to each Pre-Closing Holder, such Pre-Closing Holder's proportionate allocation (based on such Pre-Closing Holder's Closing Merger Consideration) of 0 restricted 7GC New Class A Shares, which shall be subject to the vesting and forfeiture provisions provided for in this <u>Section 2.6</u> (collectively, the "<u>Earn Out Shares</u>"):
 - (i) none of the Earn Out Shares will vest upon the occurrence of Triggering Event I (the "\$12 Earn Out Shares");
 - (ii) none of the Earn Out Shares will vest upon the occurrence of Triggering Event II (the "\$14 Earn Out Shares"); and
 - (iii) none of the Earn Out Shares will vest upon the occurrence of Triggering Event III (the "\$16 Earn Out Shares")."
- 1.9 The first sentence of Section 3.1(b) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:
- "True, correct and complete copies of the Governing Documents of each Group Company and the Company Stockholder Agreements have been provided to 7GC, in each case, as amended and in effect as of the date hereof."
- 1.10 Sections 3.4(b) through 3.4(d) of the Merger Agreement are hereby amended by deleting the following parentheticals and the language therein in their entirety:

"(including, for the avoidance of doubt, Hyros)"

- 1.11 Section 5.01(a) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:
- "(a) From and after the date of this Agreement until the earlier of the Closing or the termination of this Agreement in accordance with its terms (the "Pre-Closing Period"), the Company shall, and the Company shall cause its Subsidiaries to, except (i) as expressly required by this Agreement or any Ancillary Document, (ii) as required by applicable Law, (iii) as set forth on Section 5.1(a) of the Company Schedules, (iv) as consented to in writing by 7GC, or (v) as required to comply with COVID-19 Measures (but only to the extent reasonable and prudent in light of the business of the Group Companies and, where applicable, the circumstances giving rise to adverse changes in respect of COVID-19 or the COVID-19 Measures), operate the business of the Group Companies in the ordinary course."

- 1.12 Section 5.1(b)(ii) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:
- "(A) merge, consolidate, combine or amalgamate any Group Company with any Person, (B) purchase or otherwise acquire (whether by merging or consolidating with, purchasing any Equity Security in or a substantial portion of the assets of, or by any other manner) any business or any corporation, partnership, association or other business entity or organization or division thereof, or (C) purchase or otherwise acquire, or lease or license, any other property or assets of any other Person;"
- 1.13 Clause (C) of Section 5.1(b)(vii) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:

"[intentionally omitted];"

1.14 Clause (ii) of Section 5.3 of the Merger Agreement is hereby amended and restated in its entirety to read as follows:

"[intentionally omitted];"

1.15 Clause (y) of the second sentence of Section 5.8(a) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:

"[intentionally omitted]."

- 1.16 Section 5.8(b) of the Merger Agreement is hereby amended by replacing reach reference to "the Group Companies" with a reference to "the Group Companies".
 - 1.17 The first sentence of Section 5.17(a) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:

"The Company shall deliver to 7GC on or prior to August 31, 2023, the Closing Company Audited Financial Statements, audited in accordance with the standards of the PCAOB and containing an report of the Company's auditors qualified only as set forth on Section 5.17(a) of the Company Schedules, that is required to be included in the Registration Statement / Proxy Statement and any other filings to be made by 7GC with the SEC in connection with the transactions contemplated hereby and in the Ancillary Documents (together with the Closing Company Audited Financial Statements, the "PCAOB Financials")."

1.18 Section 5.22 of the Merger Agreement is hereby amended and restated in its entity to read as follows:

"[Intentionally omitted]."

1.19 Section 6.2(d) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:

"[intentionally omitted];"

1.20 Section 6.2(h) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:

"[intentionally omitted]; and"

- 1.21 Section 6.3(d) of the Merger Agreement is hereby amended and restated in its entirety to read as follows:
 - "(d) the Net Transaction Proceeds shall be equal to or greater than \$5,000,000."
- 1.22 <u>Section 7.1(d)</u> of the Merger Agreement is hereby amended by replacing the reference to "September 8, 2023" with "December 28, 2023", and all references in the Merger Agreement to "Termination Date" shall mean December 28, 2023.
 - $1.23 \ \underline{Section} \ 7.1(\underline{h})$ of the Merger Agreement is hereby amended and restated in its entirety to read as follows:

"[intentionally omitted];

- 1.24 Exhibit B to the Merger Agreement is hereby amended and restated in its entirety to be in the form attached hereto Exhibit B.
- 2. <u>Company Support Agreement</u>. Concurrently with the execution of this Amendment, parties to the Company Support Agreement are entering into an amended and restated support agreement in the form attached hereto as <u>Exhibit A</u> to amend and restate the Company Support Agreement in its entirety.
- 3. <u>Interpretation; No Other Amendment</u>. All capitalized terms used herein shall have the meanings assigned to them in the Merger Agreement except as otherwise provided herein or unless the context otherwise requires. For the avoidance of doubt, from and after the date of this Amendment, references in the Merger Agreement to the "Agreement" or any provision thereof shall be deemed to refer to the Merger Agreement or such provision as amended hereby unless the context otherwise requires. Except to the extent any provisions of the Merger Agreement are expressly amended by Section 1 of this Amendment, all terms and conditions of the Merger Agreement and all other documents, instruments and agreements executed thereunder, shall remain in full force and effect pursuant to the terms thereof. In the event of any inconsistency or contradiction between the terms of this Amendment and the Merger Agreement, the provisions of this Amendment shall prevail and control.
- 4. <u>General Provisions</u>. The provisions of <u>Sections 8.5</u> (Governing Law), <u>8.7</u> (Instruction; Interpretation), <u>8.10</u> (Severability), <u>8.11</u> (Counterparts; Electronic Signatures), <u>8.15</u> (Waiver of Jury Trial) and <u>8.16</u> (Jurisdiction) of the Merger Agreement apply equally to this Amendment and are hereby deemed incorporated by reference.

[Signature Page To Follow]

The Parties have caused this Amendment to be executed and delivered as of the date first written above.

BANZAI INTERNATIONAL, INC.

By: /s/ Joseph Davy

Name: Joseph Davy

Title: Chief Executive Officer

7GC & CO. HOLDINGS INC.

By: /s/ Jack Leeney

Name: Jack Leeney

Title: Chairman and Chief Executive Officer

[Amendment to Agreement and Plan of Merger and Reorganization]

Exhibit A

Amended	and Restated	Company	Support	Agreement
Amenucu	anu ixestateu	Company	Տ Ա ԻԻՍ	Agreement

AMENDED AND RESTATED COMPANY SUPPORT AGREEMENT

This AMENDED AND RESTATED COMPANY SUPPORT AGREEMENT (this "<u>Agreement</u>") is dated as of August 4, 2023, by and among 7GC & Co. Holdings Inc., a Delaware corporation ("<u>7GC</u>"), the Persons set forth on <u>Schedule I</u> hereto (each, a "<u>Stockholder</u>" and, collectively, the "<u>Stockholders</u>"), and Banzai International, Inc., a Delaware corporation (the "<u>Company</u>"). Capitalized terms used but not defined herein shall have the respective meanings ascribed to such terms in the Merger Agreement (as defined below).

RECITALS

WHEREAS, on December 8, 2022, 7GC, the Company and the Stockholders entered into that certain Company Support Agreement (the "Original Company Support Agreement");

WHEREAS, concurrently with the execution of the Original Company Support Agreement, 7GC, 7GC Merger Sub I, Inc., a Delaware corporation and an indirect wholly owned subsidiary of 7GC ("First Merger Sub"), 7GC Merger Sub II, LLC, a Delaware limited liability company and a direct wholly owned subsidiary of 7GC ("Second Merger Sub" and, together with First Merger Sub, the "Merger Subs" and each, a "Merger Sub"), and the Company entered into an Agreement and Plan of Merger and Reorganization (the "Original Merger Agreement"), pursuant to which, among other transactions, upon the terms and subject to the conditions thereof, at the Closing, First Merger Sub will merge with and into the Company (the "First Merger"), with the separate corporate existence of First Merger Sub ceasing and the Company surviving the First Merger as an indirect wholly owned subsidiary of 7GC (the "Surviving Corporation") and, promptly following the First Merger, but in any event on the same day as the First Merger and as part of the same overall transaction as the First Merger, the Surviving Corporation will merge with and into Second Merger Sub (the "Second Merger" and, together with the First Merger, the "Mergers"), with Second Merger Sub surviving the Second Merger as a wholly owned subsidiary of 7GC, and, upon the First Effective Time of the First Merger, each share of Company Stock issued and outstanding as of immediately prior to the First Effective Time shall be cancelled and automatically converted into the right to receive the consideration as described in the Original Merger Agreement (such transaction, the Mergers and the other transactions contemplated by the Original Merger Agreement, the "Transactions"), and a condition to the closing of the Transactions pursuant to the Original Merger Agreement is the completion of the Company's acquisition of Hyros Inc., a Delaware corporation ("Hyros");

WHEREAS, 7GC and the Company have decided to pursue the closing of the Transactions without completing the Company's acquisition of Hyros and requiring the completion of such acquisition as a condition to the closing of the Transactions and, concurrently with the execution of this Agreement, the Company, 7GC, and the Merger Subs are entering into that certain Amendment to Agreement Plan of Merger and Reorganization (the "Merger Agreement Amendment," and the Original Merger Agreement as amended by the Merger Agreement Amendment, the "Merger Agreement") to reflect this structural change and the corresponding changes to certain terms of the Original Merger Agreement;

WHEREAS, as of the date hereof, the Stockholders are the holders of record and "beneficial owners" (within the meaning of Rule 13d-3 of the Exchange Act) of such numbers and classes of shares of Company Stock as are indicated opposite their names on <u>Schedule I</u> attached hereto (all such Company Stock, together with any New Securities (as defined below), are referred to herein as the "<u>Subject Shares</u>");

WHEREAS, the Company and the Stockholders hereby agree to terminate, effective as of the Closing, each Company Stockholder Agreement; and

WHEREAS, pursuant to Section 2 of the Merger Agreement Amendment and as an inducement to 7GC entering into the Merger Agreement Amendment and to consummate the Transactions, 7GC, the Company and the Stockholders desire to (i) amend and restate, in its entirety, the Original Company Support Agreement by entering into this Agreement to support the closing of the Transactions pursuant to the terms of the Merger Agreement and (ii) agree to the matters as set forth herein.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing and of the mutual covenants and agreements contained herein, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

ARTICLE 1

COMPANY SUPPORT AGREEMENT; COVENANTS

1.1 Transfer of Shares. During the period commencing on the date hereof and ending on the earlier to occur of (a) the Closing, and (b) such date and time as the Merger Agreement shall be terminated in accordance with Section 7.1 of the Merger Agreement (the "Expiration Time"), each Stockholder agrees that it shall not (i) sell, assign, offer, exchange, transfer (including by operation of law), pledge, dispose of, permit to exist any Lien with respect to, or otherwise encumber (each, a "Transfer"), any of the Subject Shares or otherwise agree or commit to do any of the foregoing, (ii) deposit any Subject Shares into a voting trust or enter into a voting agreement or arrangement or grant any proxy or power of attorney with respect thereto that is inconsistent with this Agreement, (iii) enter into any contract, option or other arrangement or undertaking with respect to the direct or indirect acquisition or sale, assignment, transfer (including by operation of law) or other disposition of any Subject Shares or (iv) take any action that would have the effect of preventing, impeding, interfering with or adversely affecting such Stockholder's ability to perform its obligations under this Agreement (including pursuant to Section 1.3); provided, that the foregoing shall not prohibit the transfer of the Subject Shares, (A) if Stockholder is an individual (1) to any member of such Stockholder's immediate family, or to a trust for the benefit of Stockholder or any member of Stockholder's immediate family, the sole trustees of which are such Stockholder or any member of such Stockholder's immediate family, (2) by will, other testamentary document, under the Laws of intestacy or by virtue of Laws of descent and distribution upon the death of Stockholder, or (3) by operation of law pursuant to a qualified domestic relations order or in connection with a divorce settlement; (B) if Stockholder is an entity, to a partner, member, or Affiliate of Stockholder; or (C) if Stockholder is a trust, any beneficiary of such Stockholder or the estate of any such beneficiary; but only if, in the case of clauses (A), (B) and (C), such transferee shall concurrently execute this Agreement or a joinder agreeing to become a party to this Agreement. Any attempted transfer of Subject Shares or any interest therein in violation of this <u>Section 1.1</u> shall be null and void.

- 1.2 New Shares. In the event that, during the period commencing on the date hereof and ending at the Expiration Time, (a) any shares of Company Stock are issued to a Stockholder after the date of this Agreement pursuant to any stock dividend, stock split, recapitalization, reclassification, combination or exchange of Subject Shares or otherwise, (b) a Stockholder purchases or otherwise acquires beneficial ownership of any shares of Company Stock or (c) a Stockholder acquires the right to vote or share in the voting of any shares of Company Stock (including, without limitation, by proxy or power of attorney) (collectively the "New Securities"), then such New Securities acquired or purchased by such Stockholder shall be subject to the terms of this Agreement to the same extent as if they constituted the Subject Shares owned by such Stockholder as of the date hereof. Each of the Stockholders agrees, while this Agreement is in effect, to notify the Company promptly pursuant to the provisions of Section 3.8 of the number of New Securities acquired by such Stockholder, if any, after the date hereof.
- 1.3 <u>Agreement to Vote</u>. During the period commencing on the date hereof until the Expiration Time, each Stockholder, with respect to its, his or her Subject Shares unconditionally and irrevocably agrees that, at any meeting of the stockholders of the Company requested by the Board of Directors of the Company or undertaken as contemplated by the Transactions (or any adjournment or postponement thereof), and in any action by written consent of the stockholders of the Company (which written consent shall be delivered promptly, and in any event within twenty four (24) hours, after (x) the Registration Statement / Proxy Statement has been declared effective and (y) the Company requests such delivery), such Stockholder shall, if a meeting is held, appear at the meeting, in person or by proxy, or otherwise cause its, his or her Subject Shares to be counted as present thereat for purposes of establishing a quorum, and such Stockholder shall vote or provide consent (or cause to be voted or consented), in person or by proxy, all of its, his or her Subject Shares:
 - (a) to approve and adopt the Merger Agreement and the Transactions (including, but not limited to, the Company Preferred Conversion);
- (b) in any other circumstances upon which a consent or other approval is required under the Governing Documents of the Company or the Company Stockholder Agreements or otherwise sought with respect to, or in connection with, the Merger Agreement or the Transactions, to vote, consent or approve (or cause to be voted, consented or approved) with respect to all of such Stockholder's Subject Shares held at such time in favor thereof; and
- (c) against any 7GC Competing Transaction or any proposal, action or agreement that would impede, interfere, frustrate, delay, postpone, prevent or nullify any provision of this Agreement, the Merger Agreement or the Merger.

Each Stockholder hereby agrees that it shall not commit or agree to take any action inconsistent with the foregoing. Notwithstanding the foregoing, to the extent such Stockholder is not a director, officer or Affiliate of the Company or holder of Subject Shares representing greater than 5% of the outstanding shares of capital stock of the Company, such Stockholder shall not be obligated to vote or provide consent with respect to any of its, his or her Subject Shares or take any other action, in each case solely to the extent any such vote, consent or other action would preclude 7GC from filing with the SEC the Registration Statement / Proxy Statement on Form S-4 as contemplated by the Merger Agreement.

- 1.4 No Adverse Action; Stop Order. Each of the Stockholders agrees, while this Agreement is in effect, not to take or agree or commit to take any action that would make any representation and warranty of such Stockholder contained in this Agreement inaccurate in any material respect. Each of the Stockholders further agrees that it shall use its commercially reasonable efforts to cooperate with 7GC to effect the transactions contemplated hereby and the Transactions. In furtherance of the foregoing, the Company hereby agrees (a) to place a revocable stop order on all Subject Shares and (b) not to process any attempts by the Stockholders to Transfer any Subject Shares except in compliance with the terms of this Agreement.
- 1.5 No Challenges. Each Stockholder agrees not to commence, join in, facilitate, assist or encourage, and agrees to take all actions necessary to opt out of any class in any class action with respect to, any claim, derivative or otherwise, against 7GC, First Merger Sub, Second Merger Sub, the Company or any of their respective successors or directors (a) challenging the validity of, or seeking to enjoin the operation of, any provision of this Agreement, (b) alleging a breach of any fiduciary duty of any person in connection with the evaluation, negotiation or entry into the Merger Agreement or (c) otherwise relating to the negotiation, execution or delivery of this Agreement, the Merger Agreement or the consummation of the transactions contemplated hereby and thereby.
- 1.6 <u>Termination of Company Stockholder Agreements</u>. Each of the Company and each Stockholder hereby agrees and consents to the termination of all Company Stockholder Agreements to which such Stockholder is party, effective as of the Closing, without any further liability or obligation to such Stockholder, the Company, the Company's Subsidiaries, or 7GC. Each Stockholder agrees, confirms, and acknowledges that, after the Closing, it, he or she shall not have any of the rights or privileges provided to each such Stockholder in such applicable Company Stockholder Agreements. The termination of such Company Stockholder Agreements shall terminate the rights of the parties thereto to enforce any provisions of such agreements that expressly survive the termination of such Company Stockholder Agreements.
- 1.7 <u>Registration Rights Agreement</u>. Each of the Stockholders set forth on <u>Schedule II</u> (and any Person to whom each such Stockholder transfers its Subject Shares as permitted by the terms of this Agreement) will deliver, at the Closing, a duly executed copy of the Registration Rights Agreement, substantially in the form attached as Exhibit B to the Merger Agreement.
- 1.8 <u>Lock-up Agreements</u>. Each of the Stockholders set forth on <u>Schedule III</u> (and any Person to whom each such Stockholder transfers its Subject Shares as permitted by the terms of this Agreement) will deliver, at the Closing, a duly executed copy of the Lock-up Agreement, substantially in the form attached as Exhibit C to the Merger Agreement.
- 1.9 <u>Appraisal and Dissenters' Rights</u>. Each Stockholder hereby irrevocably and unconditionally waives, and agrees not to assert or perfect, any rights of appraisal or other similar rights to dissent (including any notice requirements related thereto) with respect to the Merger, the Merger Agreement or any of the Transactions that Stockholder may have by virtue of ownership of Subject Shares (including all rights under Section 262 of the DGCL).

1.10 Exclusivity. Unless this Agreement shall have been terminated in accordance with Section 3.1, each Stockholder agrees not to, and shall cause its, his or her Affiliates or Representatives not to, (a) accept, initiate, respond to, encourage, entertain, solicit, negotiate, provide information with respect to or discuss other offers for an Acquisition Proposal; (b) furnish or disclose any non-public information to any Person in connection with, or that could reasonably be expected to lead to, an Acquisition Proposal; (c) enter into any Contract regarding an Acquisition Proposal; (d) prepare or take any steps in connection with a public offering of any Equity Securities of any Group Company (or any successor to or parent company of any Group Company); or (e) otherwise cooperate in any way with, or assist or participate in, or facilitate or encourage any effort or attempt by any Person to do or seek to do any of the foregoing or seek to circumvent this Section 1.10 or further an Acquisition Proposal. Each Stockholder agrees to (A) notify the Company and 7GC promptly upon receipt (and in any event within forty-eight (48) hours after receipt) of any Acquisition Proposal, and to describe the terms and conditions of any such Acquisition Proposal in reasonable detail (including the identity of the Persons making such Acquisition Proposal), (B) keep the Company and 7GC fully informed on a current basis of any material modifications to such offer or information and (C) not (and shall cause its Affiliates and Representatives not to) conduct any further discussions with, provide any information to, or enter into negotiations with such Persons. Each Stockholder shall immediately cease and cause to be terminated any discussions or negotiations with any Persons (other than 7GC and its Representatives) that may be ongoing with respect to an Acquisition Proposal and terminate any such Person's and such Person's Representative's access to any electronic data room. Each Stockholder shall not release any third party from, or wa

1.11 Consent to Disclosure. Each Stockholder hereby consents to the publication and disclosure in the Registration Statement / Proxy Statement (and, as and to the extent otherwise required by applicable securities Laws or the SEC or any other securities authorities, any other documents or communications provided by the Company or 7GC to any Governmental Entity or to securityholders of 7GC) of such Stockholder's identity and beneficial ownership of its Subject Shares and the nature of such Stockholder's commitments, arrangements and understandings under and relating to this Agreement and, if deemed appropriate by the Company or 7GC, a copy of this Agreement. Each Stockholder will promptly provide any information reasonably requested by the Company or 7GC in connection with the first sentence of this Section 1.11 or for any regulatory application or filing made or approval sought in connection with the Transactions (including filings with the SEC).

REPRESENTATIONS AND WARRANTIES

2.1 <u>Representations and Warranties of the Stockholders</u>. Each Stockholder represents and warrants as of the date hereof to 7GC and the Company (solely with respect to itself, himself or herself and not with respect to any other Stockholder) as follows:

(a) <u>Organization</u>; <u>Due Authorization</u>. If such Stockholder is not an individual, it is duly organized, validly existing and in good standing under the Laws of the jurisdiction in which it is incorporated, formed, organized or constituted, and the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby are within such Stockholder's corporate, limited liability company or organizational powers and have been duly authorized by all necessary corporate, limited liability company or organizational actions on the part of such Stockholder. If such Stockholder is an individual, such Stockholder has full legal capacity, right and authority to execute and deliver this Agreement and to perform his or her obligations hereunder. This Agreement has been duly executed and delivered by such Stockholder and, assuming due authorization, execution and delivery by the other parties to this Agreement, this Agreement constitutes a legally valid and binding obligation of such Stockholder, enforceable against such Stockholder in accordance with the terms hereof (except as enforceability may be limited by bankruptcy Laws, other similar Laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies). If any Stockholder is an individual and is married, and any of the Subject Shares of such Stockholder constitute community property or spousal approval is otherwise necessary for this Agreement to be legal, binding and enforceable with respect to such Stockholder, such Stockholder's spouse has delivered with this Agreement a Spousal Consent in the form attached hereto as <u>Exhibit A</u> and this Agreement has been duly authorized, executed and delivered by, and constitutes the legal, valid and binding obligation of, such Stockholder's spouse, enforceable against such Stockholder's spouse in accordance with its terms. If this Agreement is being executed in a representative or fiduciary capacity, the Perso

(b) Ownership. Such Stockholder is the record and beneficial owner (as defined in the Securities Act) of, and has good, valid and marketable title to, all of such Stockholder's Subject Shares, and there exist no Liens or any other limitation or restriction (including any restriction on the right to vote, sell or otherwise dispose of such Subject Shares) affecting any such Subject Shares, other than Liens pursuant to (i) this Agreement, (ii) the Governing Documents of the Company, (iii) the Company Stockholder Agreements, (iv) the Merger Agreement, or (v) any applicable securities Laws. Such Stockholder's Subject Shares are the only Equity Securities in the Company owned of record or beneficially by such Stockholder on the date of this Agreement, and none of such Stockholder's Subject Shares are subject to any proxy, voting trust or other agreement or arrangement with respect to the voting of such Subject Shares, except as provided hereunder and under the Company Stockholder Agreements. Such Stockholder has full and sole voting power (including the right to control such vote as contemplated herein), full power of disposition and full power to agree to all of the matters applicable to such Stockholder set forth in this Agreement, in each case, with respect to such Stockholder's Subject Shares (including, without limitation, by proxy or power of attorney). Such Stockholder does not hold or own any rights to acquire (directly or indirectly) any Equity Securities of the Company or any Equity Securities convertible into, or which can be exchanged for, Equity Securities of the Company. Such Stockholder has not granted a proxy or power of attorney with respect to any of the Stockholder's Subject Shares that is inconsistent with the Stockholder's obligations pursuant to this Agreement. There are no claims for finder's fees or brokerage commissions or other like payments in connection with this Agreement or the transactions contemplated hereby payable by such Stockholder pursuant to arrangements made by such Stockholder.

- (c) No Conflicts. The execution and delivery of this Agreement by such Stockholder does not, and the performance by such Stockholder of his, her or its obligations hereunder and the compliance by such Stockholder of his, her or its obligations hereunder do not and will not, (i) if such Stockholder is not an individual, contravene, conflict with or result in a violation of the organizational documents of such Stockholder or (ii) require any consent, authorization or approval that has not been given or other action that has not been taken by any Person (including under any Contract binding upon such Stockholder or such Stockholder's Subject Shares), in each case, to the extent such consent, approval or other action would prevent, enjoin or materially delay the performance by such Stockholder of its, his or her obligations under this Agreement, (iii) conflict with or violate any Law, or (iv) result in the creation of a Lien on any of the Subject Shares. No consent, approval, order or authorization of, or registration, declaration or filing with, any Governmental Entity or any other person is required by or with respect to such Stockholder in connection with the execution and delivery of this Agreement or the consummation by such Stockholder of the transactions contemplated hereby. If such Stockholder is an individual, no consent of such Stockholder's spouse is necessary under any "community property" or other Laws in order for such Stockholder to enter into and perform its obligations under this Agreement.
- (d) <u>Litigation</u>. There are no Proceedings pending against such Stockholder, or to the knowledge of such Stockholder threatened against such Stockholder, before (or, in the case of threatened Proceedings, that would be before) any arbitrator or any Governmental Entity, which in any manner challenges, seeks to prevent, enjoin or materially delay or could have the effect of preventing, enjoining or materially delaying the performance by such Stockholder of its, his or her obligations under this Agreement.
- (e) <u>Adequate Information</u>. Such Stockholder is a sophisticated stockholder and has adequate information concerning the business and financial condition of 7GC and the Company to make an informed decision regarding this Agreement, the Transactions and the transactions contemplated hereby and has independently and without reliance upon 7GC or the Company and based on such information as such Stockholder has deemed appropriate, made its, his or her own analysis and decision to enter into this Agreement. Such Stockholder acknowledges that 7GC and the Company have not made and do not make any representation or warranty, whether express or implied, of any kind or character except as expressly set forth in this Agreement. Such Stockholder acknowledges that the agreements contained herein with respect to the Subject Shares held by such Stockholder are irrevocable.
- (f) <u>Acknowledgment</u>. Such Stockholder understands and acknowledges that each of 7GC and the Company is entering into the Merger Agreement in reliance upon such Stockholder's execution and delivery of this Agreement.
- (g) No Inconsistent Agreements. Such Stockholder hereby covenants and agrees that, except for this Agreement, such Stockholder (a) has not entered into, nor will enter into at any time while this Agreement remains in effect, any voting agreement or voting trust with respect to such Stockholder's Subject Shares inconsistent with such Stockholder's obligations pursuant to this Agreement, (b) has not granted, nor will grant at any time while this Agreement remains in effect, a proxy, consent or power of attorney with respect to such Stockholder's Subject Shares and (c) has not entered into any agreement or knowingly taken any action (nor will enter into any agreement or knowingly take any action) that would make any representation or warranty of such Stockholder contained herein untrue or incorrect in any material respect or have the effect of preventing such Stockholder from performing any of its material obligations under this Agreement.

ARTICLE 3

MISCELLANEOUS

- 3.1 <u>Termination</u>. This Agreement and all of its provisions shall automatically terminate upon the earliest of (a) the Expiration Time and (b) as to each Stockholder, the written agreement of 7GC, the Company and such Stockholder. Upon such termination of this Agreement, all obligations of the parties under this Agreement will terminate, without any liability or other obligation on the part of any party hereto to any Person in respect hereof or the transactions contemplated hereby, and no party hereto shall have any claim against another (and no Person shall have any rights against such party), whether under contract, tort or otherwise, with respect to the subject matter hereof; <u>provided</u>, <u>however</u>, that the termination of this Agreement shall not relieve any party hereto from liability arising in respect of any breach of this Agreement prior to such termination. This <u>Article III</u> shall survive the termination of this Agreement.
- 3.2 <u>Governing Law</u>. This Agreement and all related Proceedings shall be governed by and construed in accordance with the internal Laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Law of any jurisdiction other than the State of Delaware.

3.3 CONSENT TO JURISDICTION AND SERVICE OF PROCESS; WAIVER OF JURY TRIAL.

(a) ANY PROCEEDING BASED UPON, ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY MUST BE BROUGHT IN THE COURT OF CHANCERY OF THE STATE OF DELAWARE (OR, TO THE EXTENT SUCH COURT DOES NOT HAVE SUBJECT MATTER JURISDICTION, THE SUPERIOR COURT OF THE STATE OF DELAWARE), OR, IF IT HAS OR CAN ACQUIRE JURISDICTION, IN THE UNITED STATES DISTRICT COURT FOR THE DISTRICT OF DELAWARE, AND EACH OF THE PARTIES IRREVOCABLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF EACH SUCH COURT IN ANY SUCH PROCEEDING, WAIVES ANY OBJECTION IT MAY NOW OR HEREAFTER HAVE TO PERSONAL JURISDICTION, VENUE OR TO CONVENIENCE OF FORUM, AGREES THAT ALL CLAIMS IN RESPECT OF THE PROCEEDING SHALL BE HEARD AND DETERMINED ONLY IN ANY SUCH COURT, AND AGREES NOT TO BRING ANY PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY IN ANY OTHER COURT. NOTHING HEREIN CONTAINED SHALL BE DEEMED TO AFFECT THE RIGHT OF ANY PARTY TO SERVE PROCESS IN ANY MANNER PERMITTED BY LAW OR TO COMMENCE LEGAL PROCEEDINGS OR OTHERWISE PROCEED AGAINST ANY OTHER PARTY IN ANY OTHER JURISDICTION, IN EACH CASE, TO ENFORCE JUDGMENTS OBTAINED IN ANY PROCEEDING BROUGHT PURSUANT TO THIS <u>SECTION 3.3(a)</u>.

(b) WAIVER OF TRIAL BY JURY. THE PARTIES EACH HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION, OR CAUSE OF ACTION (I) ARISING UNDER THIS AGREEMENT OR (II) IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES IN RESPECT OF THIS AGREEMENT OR ANY OF THE TRANSACTIONS RELATED HERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER IN CONTRACT, TORT, EQUITY, OR OTHERWISE. THE PARTIES EACH HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION, OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY AND THAT THE PARTIES MAY FILE AN ORIGINAL COUNTERPART OF A COPY OF THIS AGREEMENT WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE PARTIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

- 3.4 <u>Assignment</u>. This Agreement may not be assigned by any party (whether by operation of Law or otherwise) without the prior written consent of the other parties hereto. Any attempted assignment of this Agreement not in accordance with the terms of this <u>Section 3.4</u> shall be void, *ab initio*.
- 3.5 <u>Specific Performance</u>. The parties agree that irreparable damage for which monetary damages, even if available, would not be an adequate remedy, would occur in the event that the parties do not perform their respective obligations under the provisions of this Agreement (including failing to take such actions as are required of them hereunder to consummate the Transactions) in accordance with their specific terms or otherwise breach such provisions. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions, specific performance and other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, including the Stockholder's obligations to vote its Subject Shares as provided in this Agreement, in each case without posting a bond or undertaking and without proof of damages and this being in addition to any other remedy to which they are entitled at Law or in equity. Each of the parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief when expressly available pursuant to the terms of this Agreement on the basis that the other parties have an adequate remedy at Law or an award of specific performance is not an appropriate remedy for any reason at Law or equity.
- 3.6 <u>Amendment; Waiver</u>. This Agreement may be amended or modified only by a written agreement executed and delivered by 7GC, the Company and the Stockholders. This Agreement may not be modified or amended except as provided in the immediately preceding sentence and any purported amendment by any party or parties effected in a manner which does not comply with this <u>Section 3.6</u> shall be void, *ab initio*.
- 3.7 Severability. Whenever possible, each provision of this Agreement will be interpreted in such a manner as to be effective and valid under applicable Law, but if any term or other provision of this Agreement is held to be invalid, illegal or unenforceable under applicable Law, all other provisions of this Agreement shall remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision of this Agreement is invalid, illegal or unenforceable under applicable Law, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby are consummated as originally contemplated to the greatest extent possible.

3.8 <u>Notices</u>. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given) when delivered in person, when delivered by e-mail (having obtained electronic delivery confirmation thereof), or when sent by registered or certified mail (postage prepaid, return receipt requested) (upon receipt thereof) to the other parties as follows:

If to 7GC:

c/o 7GC & Co. Holdings LLC 388 Market Street, Suite 1300 San Francisco, CA 94111 Attention:Jack Leeney Chris Walsh

E-mail: jack@7GC.co chris@7GC.co

with a copy (which shall not constitute notice) to:

Sidley Austin LLP 1999 Avenue of the Stars 17th Floor Los Angeles, CA 90067 Attention:Joshua G. DuClos E-mail: jduclos@sidley.com

Sidley Austin LLP
One South Dearborn Street
Chicago, IL 60603
Attention:Michael P. Heinz and Matthew D. Stoker
E-mail: mheinz@sidley.com and mstoker@sidley.com

If to the Company:

Banzai International, Inc. 435 Ericksen Ave, Suite 250 Bainbridge Island, WA 98110 Attention:Joseph Davy E-mail: joe@banzai.io

with a copy (which shall not constitute notice) to:

Cooley LLP 1700 Seventh Avenue, Suite 1900 Seattle, WA 98101-1355 Attention:Sonya Erickson E-mail: serickson@cooley.com

If to a Stockholder:

To such Stockholder's address set forth on Stockholder's signature page hereto.

or to such other address as the party to whom notice is given may have previously furnished to the others in writing in the manner set forth above.

- 3.9 <u>Further Assurances</u>. Each Stockholder shall execute and deliver, or cause to be delivered, such additional documents, and take, or cause to be taken, all such further actions and do, or cause to be done, all things reasonably necessary (including under applicable Laws), or reasonably requested by 7GC or the Company, to effect the actions and consummate the Merger and the other transactions contemplated by this Agreement and the Merger Agreement (including the Transactions), in each case, on the terms and subject to the conditions set forth therein and herein, as applicable.
- 3.10 Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed to be an original, but all of which shall constitute one and the same agreement. The words "execution," "signed," "signature," and words of like import in this Agreement shall include images of manually executed signatures transmitted by facsimile or other electronic format (including, "pdf", "tif" or "jpg") and other electronic signatures (including, DocuSign and AdobeSign). The use of electronic signatures and electronic records (including, any contract or other record created, generated, sent, communicated, received, or stored by electronic means) shall be of the same legal effect, validity and enforceability as a manually executed signature or use of a paper-based record-keeping system to the fullest extent permitted by applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the Delaware Uniform Electronic Transactions Act and any other applicable law. Minor variations in the form of the signature page, including footers from earlier versions of this Agreement, shall be disregarded in determining the party's intent or the effectiveness of such signature.
- 3.11 Entire Agreement; Effect of this Agreement. This Agreement constitutes the full and entire agreement among the parties hereto with respect to the subject matter hereof and supersedes all other prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof. For the avoidance of doubt and in furtherance of the foregoing, upon the execution and delivery of this Agreement, by the Company, Stockholders and 7GC, the Original Support Agreement shall automatically terminate and be of no further force and effect and the Original Company Support Agreement is hereby amended and restated in its entirety as set forth in this Agreement.

[Remainder of Page Intentionally Left Blank]

7GC & CO. HOLDINGS INC.

By:	
Name:	Jack Leeney
Title:	Chairman and Chief Executive Officer

[SIGNATURE PAGE TO AMENDED AND RESTATED COMPANY SUPPORT AGREEMENT]

BANZAI INTERNATIONAL, INC.

ву:	·
Name:	Joseph P. Davy
Title:	Chief Executive Officer

[SIGNATURE PAGE TO AMENDED AND RESTATED COMPANY SUPPORT AGREEMENT]

Sheila Linteau, as Administrator of the estate of Roland A. Linteau III, Deceased

Address: ****
Email: ****

[SIGNATURE PAGE TO AMENDED AND RESTATED COMPANY SUPPORT AGREEMENT]

Joseph P. Davy

Address: ****
Email: ****

[SIGNATURE PAGE TO AMENDED AND RESTATED COMPANY SUPPORT AGREEMENT]

A-15

DNX PARTNERS III, LP

By: DNX Venture Partners III, LP,

its General Partner

By: DNX III, LLC, its General Partner

By:

Name: Mitch Kitamura
Title: Authorized Signatory

DNX PARTNERS JAPAN III, LP

By: DNX Venture Partners III, LP,

its General Partner

By: DNX III, LLC, its General Partner

By:

Name: Mitch Kitamura
Title: Authorized Signatory

DNX PARTNERS S-III, LP

By: DNX Partners S3, LLC

its General Partner

By:

Name: Mitch Kitamura
Title: Authorized Signatory

[Signature Page to Amended and Restated Company Support Agreement]

SCHEDULE I

Stockholder Subject Shares

Stockholder	Class of Shares	Number of Shares
DNX Partners III, LP	Series A-1 Preferred Stock	1,014,166
DNX Partners Japan III, LP	Series A-1 Preferred Stock	350,266
DNX Partners S-III, LP	Series A-1 Preferred Stock	7,545
Joseph P. Davy	Class B Common Stock	3,760,000
The Estate of Roland A. Linteau III	Class B Common Stock	2,560,000
	A-17	

SCHEDULE II

Registration Rights Agreement Signatories

- 1. Joseph Davy
- 2. Sheila Linteau on behalf of the estate of Roland Linteau
- 3. DNX Partners III, LP
- 4. DNX Partners Japan III, LP
- 5. DNX Partners S-III, LP

SCHEDULE III

Lock-up Agreement Signatories

- 1. Joseph Davy
- 2. Sheila Linteau on behalf of the estate of Roland Linteau
- 3. DNX Partners III, LP
- 4. DNX Partners Japan III, LP
- 5. DNX Partners S-III, LP

EXHIBIT A

Form of Spousal Consent

I,	, spouse of	("Stockholder"), acknowledge that I have read the Amended and Restated Company
		. Holdings Inc., a Delaware corporation (" <u>7GC</u> "), and Banzai International, Inc., a Delaware "), and that I know the contents of the Agreement. Capitalized terms used but not defined
	neanings assigned to them in the Agreem	
including any interest irrevocably subject to	that I might have therein. I understand the Agreement and the other agreements	regarding Company Stock (as defined in the Merger Agreement) that my spouse may own, and agree that my interest, if any, in any Company Stock subject to the Agreement shall be seferred to therein. I further understand and agree that any community property interest that the other agreement and the other agreements referred to therein.
acknowledge, deliver, necessary, incidental, Company Stock or Ne without further signatu of any and all such Co that the foregoing pow	file and record all documents which meconvenient or otherwise), any and all of we Securities (as defined in the Agreement, consent or knowledge, to exercise a mpany Stock and New Securities), with er of attorney and proxy is coupled with	my true and lawful attorney and proxy in my name, place and stead to sign, make, execute, may be required, and to manage, vote, act and make all decisions with respect to (whether Company Stock in which I now have or hereafter acquire any interest and in any and all lent) now or hereafter held of record by Stockholder (including but not limited to, the right, amendments and modifications of, and to terminate, the foregoing agreements and to dispose all powers I would possess if personally present, it being expressly understood and intended an interest; and this power of attorney is a durable power of attorney and will not be affected in of marriage and this proxy will not terminate without the consent of Stockholder, 7GC and
	ith respect to this consent. I have either s	s contained in the Agreement are complex and that I am free to seek independent professional sought such guidance or counsel or determined after carefully reviewing the Agreement that I α
		Name:
		Date:
		A-20

Exhibit B

Form of Registration Rights Agreement

AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

This AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT (this "Agreement"), dated as of [●], 2023, is made and entered into by and among [Banzai Holdings, Inc.] (formerly known as 7GC & Co. Holdings Inc.), a Delaware corporation (the "Company"), 7GC & Co. Holdings LLC, a Delaware limited liability company (the "Sponsor"), each of the undersigned parties that holds Founder Shares (as defined below) and is identified as an "Other Pre-IPO Holder" on the signature pages hereto (collectively, with the Sponsor, the "Existing Holders"), and the undersigned parties identified as "New Holders" on the signature pages hereto (collectively, the "New Holders") (each of the foregoing parties (other than the Company) and any person or entity who hereafter becomes a party to this Agreement pursuant to Section 5.2 of this Agreement, a "Holder" and collectively, the "Holders"). Capitalized terms used but not otherwise defined in this Agreement shall have the meanings ascribed thereto in the Merger Agreement (as defined below).

RECITALS

WHEREAS, the Company and the Sponsor entered into that certain Securities Subscription Agreement, dated as of September 18, 2020, pursuant to which the Sponsor purchased an aggregate of 5,031,250 shares of the Company's Class B common stock, par value \$0.0001 per share ("*Class B Common Stock*"), which were issued in a private placement prior to the closing of the Company's initial public offering;

WHEREAS, on December 1, 2020, the Sponsor transferred 25,000 shares of Class B Common Stock held by the Sponsor to each of Kent Schofield, Tripp Jones, Patrick Eggen and Courtney Robinson (collectively, the "*Company Directors*");

WHEREAS, in December 2020, the Company effected a stock dividend of approximately 0.143 of a share for each share of Class B Common Stock outstanding, resulting in an aggregate of 5,750,000 shares of Class B Common Stock outstanding (the outstanding shares of Class B Common Stock held by each of the Sponsor and the Company Directors being referred to herein as the "*Founder Shares*");

WHEREAS, upon the consummation of the transactions (the "*Closing*") contemplated by the Merger Agreement (as defined below), the Founder Shares shall be automatically converted into shares of the Company's Class A common stock, par value \$0.0001 per share ("*Class A Common Stock*"), on a one-for-one basis, subject to adjustment;

WHEREAS, in order to finance the Company's transaction costs in connection with the transactions contemplated by the Merger Agreement, the Sponsor or an affiliate of the Sponsor or certain of the Company's officers and directors may, but are not obligated to, loan to the Company funds as the Company may require, of which up to \$1,500,000 of such loans may be convertible into up to an additional 1,500,000 warrants at a price of \$1.00 per warrant (the "Working Capital Warrants");

WHEREAS, the Company entered into that certain Agreement and Plan of Merger and Reorganization, dated as of December 8, 2022 (the "*Merger Agreement*"), by and among the Company, 7GC Merger Sub I, Inc., a Delaware corporation and an indirect wholly owned subsidiary of the Company ("*First Merger Sub*"), 7GC Merger Sub II, LLC, a Delaware limited liability company and a direct wholly owned subsidiary of the Company ("*Second Merger Sub*" and, together with First Merger Sub, the "*Merger Subs*" and each, a "*Merger Sub*"), and Banzai International, Inc., a Delaware corporation ("*Banzai*"), pursuant to which, through a series of mergers at the Closing with the Merger Subs, Banzai will be merged with and into Second Merger Sub, with Second Merger Sub surviving the mergers and remaining a wholly-owned subsidiary of the Company;

WHEREAS, pursuant to the transactions contemplated by the Merger Agreement and subject to the terms and conditions set forth therein, upon the Closing, the New Holders will receive shares of the Company's Class A Common Stock and/or Class B Common Stock;

WHEREAS, each of the Company and the Existing Holders is a party to that certain Registration Rights Agreement, dated December 22, 2020 (the "*Existing Registration Rights Agreement*"), pursuant to which the Company granted the Existing Holders certain registration rights with respect to certain securities of the Company, as set forth therein; and

WHEREAS, upon the Closing, each party to the Existing Registration Rights Agreement desires to amend and restate the Existing Registration Rights Agreement in its entirety as set forth herein, and the Company and the Holders desire to enter into this Agreement, pursuant to which the Company shall grant the Holders certain registration rights with respect to the Registrable Securities (as defined below) on the terms and conditions set forth in this Agreement.

NOW, **THEREFORE**, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

- 1.1 <u>Definitions</u>. The terms defined in this <u>Article I</u> shall, for all purposes of this Agreement, have the respective meanings set forth below:
- "Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Chief Executive Officer or principal financial officer of the Company, after consultation with counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, declared effective, or used, as the case may be, and (iii) the Company has a bona fide business purpose for not making such information public.
 - "Agreement" shall have the meaning given in the Preamble.
 - "Banzai" shall have the meaning given in the Recitals hereto.
 - "Board" shall mean the Board of Directors of the Company.
- "business day" shall mean a day, other than a Saturday or Sunday, on which commercial banks in San Francisco, California or Salt Lake City, Utah are open for the general transaction of business.
 - "Closing" shall have the meaning given in the Recitals hereto.
 - "Commission" shall mean the U.S. Securities and Exchange Commission.

- "Common Stock" shall mean the common stock, par value \$0.0001 per share, of the Company (which, for the avoidance of doubt, is comprised of shares designated as "Class A" and "Class B" common stock).
 - "Company" shall have the meaning given in the Preamble.
 - "Company Underwritten Demand Notice" shall have the meaning given in subsection 2.1.3.
 - "Demanding Holder" shall have the meaning given in subsection 2.1.3.
 - "Effectiveness Deadline" shall have the meaning given in subsection 2.1.1.
 - "Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.
 - "Existing Holders" shall have the meaning given in the Preamble.
 - "Existing Registration Rights Agreement" shall have the meaning given in the Recitals hereto.
 - "First Merger Sub" shall have the meaning given in the Recitals hereto.
- "Form S-1 Registration Statement" shall mean a Registration Statement on Form S-1 allowing for the delayed or continuous offering and sale of the Registrable Securities pursuant to Rule 415.
- "Form S-3 Shelf" shall mean a Registration Statement on Form S-3 allowing for the delayed or continuous offering and sale of the Registrable Securities pursuant to Rule 415.
- "Founder Shares" shall have the meaning given in the Recitals hereto and shall, for the avoidance of doubt, be deemed to include the shares of Class A Common Stock issuable upon conversion thereof.
- "Founder Shares Lock-Up Period" shall mean, with respect to the Founder Shares, from the date hereof until the earlier of (A) 180 days after the date hereof; and (B) the date on which the Company completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of the Company's stockholders having the right to exchange their shares of Common Stock for cash, securities or other property.
- "GEM Agreement" shall mean that certain Share Purchase Agreement, dated as of May 27, 2022, by and between the Company and GEM Global Yield LLC SCS and GEM Yield Bahamas Limited.
 - "Holder" and "Holders" shall have the meanings given in the Preamble.
- "*Insider Letter*" shall mean that certain letter agreement, dated as of December 22, 2020, by and among the Company, the Sponsor and each of the Company's officers and directors signatory thereto.
 - "Lock-Up Period" shall mean the Founder Shares Lock-Up Period.
 - "Maximum Number of Securities" shall have the meaning given in subsection 2.1.5.
 - "Merger Agreement" shall have the meaning given in the Recitals hereto.

- "Merger Sub" and "Merger Subs" shall have the meanings given in the Recitals hereto.
- "*Misstatement*" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus, or necessary to make the statements in a Registration Statement or Prospectus (in the case of the Prospectus, in the light of the circumstances under which they were made) not misleading.
 - "New Holders" shall have the meaning given in the Preamble.
- "*Permitted Transferees*" shall mean any person or entity (i) to whom a Holder of Registrable Securities is permitted to transfer such Registrable Securities prior to the expiration of the Lock-Up Period, under the Insider Letter, this Agreement and any other applicable agreement between such Holder and the Company, and to any transferee thereafter, and (ii) who agrees to become bound by the transfer restrictions set forth in this Agreement.
 - "Piggyback Registration" shall have the meaning given in subsection 2.2.1.
 - "Pro Rata" shall have the meaning given in subsection 2.1.5.
- "*Prospectus*" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.
- "Registrable Security" shall mean (a) the Founder Shares (including, for the avoidance of doubt, the shares of Class A Common Stock issued or issuable upon the conversion of any Founder Shares), (b) any issued and outstanding share of Common Stock or any other equity security (including the shares of Common Stock issued or issuable upon the exercise of any other equity security) of the Company held by an Existing Holder as of the date of this Agreement, (c) any shares of Common Stock issued or issuable upon exercise of the Working Capital Warrants, (d) any outstanding shares of Common Stock or any other equity security of the Company held by a New Holder as of the date of this Agreement (including shares transferred to a Permitted Transferee and the shares of Common Stock issued or issuable upon the exercise of any such other equity security) and (e) any other equity security of the Company issued or issuable with respect to any such share of the Common Stock described in the foregoing clauses (a) through (e) by way of a stock dividend or stock split or in connection with a combination of shares, distribution, recapitalization, merger, consolidation or reorganization or other similar event; provided, however, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities upon the earliest to occur of: (i) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (ii) such securities shall have been otherwise transferred, new certificates or book entry positions for such securities not bearing a legend restricting further transfer shall have been otherwise transferred, new certificates or book entry positions for such securities not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not

"*Registration*" shall mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

- "Registration Expenses" shall mean the out-of-pocket expenses of a Registration, including, without limitation, the following:
- (A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Common Stock is then listed;
- (B) fees and expenses of compliance with securities or blue sky laws (including reasonable and customary fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);
 - (C) printing, messenger, telephone and delivery expenses;
 - (D) reasonable fees and disbursements of counsel for the Company;
- (E) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration; and
- (F) reasonable fees and expenses of one (1) legal counsel selected by the majority-in-interest of the Demanding Holders initiating an Underwritten Demand to be registered for offer and sale in the applicable Registration.
- "*Registration Statement*" shall mean any registration statement under the Securities Act that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.
 - "Requesting Holders" shall have the meaning given in subsection 2.1.3.
 - "Restricted Securities" shall have the meaning given in subsection 3.6.1.
- "Rule 144" shall mean Rule 144 promulgated under the Securities Act (together with any successor rule promulgated thereafter by the Commission).
 - "Rule 415" shall have the meaning given in subsection 2.1.1.
 - "Second Merger Sub" shall have the meaning given in the Recitals hereto.
 - "Securities Act" shall mean the Securities Act of 1933, as amended from time to time.
 - "Sponsor" shall have the meaning given in the Preamble.
- "*Underwriter*" shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.
 - "Underwritten Demand" shall have the meaning given in subsection 2.1.3.
 - "Underwritten Demand Notice" shall have the meaning given in subsection 2.1.3.
- "Underwritten Registration" or "Underwritten Offering" shall mean a Registration in which securities of the Company are sold to an Underwritter in a firm commitment underwriting for distribution to the public, including an offering and/or sale of Registrable Securities by any Holder in a block trade or on an underwritten basis (whether firm commitment or otherwise) without substantial marketing efforts prior to pricing, including, without limitation, a same day trade, overnight trade or similar transaction, but excluding a variable price reoffer.
 - "Working Capital Warrants" shall have the meaning given in the Recitals hereto.

ARTICLE II REGISTRATIONS

2.1 Shelf Registration.

2.1.1 Initial Registration. The Company shall use its reasonable best efforts to, as promptly as reasonably practicable, but in no event later than thirty (30) days after the Closing (the "Filing Deadline"), file a Form S-1 Registration Statement under the Securities Act to permit the public resale of all the Registrable Securities held by the Holders (and certain other outstanding equity securities of the Company) from time to time as permitted by Rule 415 under the Securities Act (or any successor or similar provision adopted by the Commission then in effect) ("Rule 415") on the terms and conditions specified in this subsection 2.1.1 and shall use its reasonable best efforts to cause such Registration Statement to be declared effective as promptly as reasonably practicable after the initial filing thereof, but in no event later than sixty (60) business days following the Filing Deadline (the "Effectiveness Deadline"); provided, that the Effectiveness Deadline shall be extended to one hundred and twenty (120) days after the Filing Deadline if the Registration Statement is reviewed by, and receives comments from, the Commission. The Registration Statement filed with the Commission pursuant to this subsection 2.1.1 shall be on a Form S-1 Registration Statement or such other form of registration statement as is then available to effect a registration for resale of such Registrable Securities, covering such Registrable Securities, and shall contain a Prospectus in such form as to permit any Holder to sell such Registrable Securities pursuant to Rule 415 at any time beginning on the effective date for such Registration Statement. A Registration Statement filed pursuant to this subsection 2.1.1 shall provide for the resale pursuant to any method or combination of methods legally available to, and reasonably requested prior to effectiveness by, the Holders. The Company shall use its reasonable best efforts to cause a Registration Statement filed pursuant to this subsection 2.1.1 to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, that another Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities. When effective, a Registration Statement filed pursuant to this subsection 2.1.1 (including the documents incorporated therein by reference) will comply as to form in all material respects with all applicable requirements of the Securities Act and the Exchange Act and will not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading (in the case of any Prospectus contained in such Registration Statement, in the light of the circumstances under which such statement is made).

2.1.2 Form S-3 Shelf. The Company shall use its commercially reasonable efforts to convert the Form S-1 Registration Statement to a Form S-3 Shelf as soon as practicable after the Company is eligible to use a Form S-3 Shelf. If the Company files a Form S-3 Shelf and thereafter the Company becomes ineligible to use Form S-3 for secondary sales, the Company shall use its reasonable best efforts to file a Form S-1 Registration Statement as promptly as reasonably practicable to replace the shelf registration statement that is a Form S-3 Shelf and have the Form S-1 Registration Statement declared effective as promptly as reasonably practicable and to cause such Form S-1 Registration Statement to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, that another Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities.

2.1.3 Underwritten Offering. At any time and from time to time following the effectiveness of the Registration Statement required by subsections 2.1.1 or 2.1.2, any Holder may request to sell all or a portion of their Registrable Securities (a "Demanding Holder") in an underwritten offering that is registered pursuant to such Registration Statement (an "Underwritten Demand"), provided that such Holder(s) (a) reasonably expect aggregate gross proceeds in excess of \$50,000,000 from such Underwritten Offering or (b) reasonably expects to sell all of the Registrable Securities held by such Holder in such Underwritten Offering but in no event less than \$10,000,000 in aggregate gross proceeds. All requests for an Underwritten Offering shall be made by giving written notice to the Company (the "Underwritten Demand Notice"). Each Underwritten Demand Notice shall specify the approximate number of Registrable Securities proposed to be sold in the Underwritten Offering and the expected price range (net of underwriting discounts and commissions) of such Underwritten Offering. Within five (5) business days after receipt of any Underwritten Demand Notice, the Company shall give written notice of such requested Underwritten Offering (the "Company Underwritten Demand Notice") to all other Holders of Registrable Securities (the "Requesting Holders") and, subject to reductions consistent with the Pro Rata calculations in subsection 2.1.5, shall include in such Underwritten Offering all Registrable Securities with respect to which the Company has received written requests for inclusion therein, within five (5) days after sending the Company Underwritten Demand Notice. The Company shall enter into an underwriting agreement in a form as is customary in Underwritten Offerings of securities by the Company with the managing Underwriter or Underwriters selected by the initiating Demanding Holders with the written consent of the Company (such consent not to be unreasonably withheld, conditioned or delayed) and shall take all such other reasonable actions as are requested by the managing Underwriter or Underwriters in order to expedite or facilitate the disposition of such Registrable Securities. In connection with any Underwritten Offering contemplated by this <u>subsection 2.1.3</u>, subject to <u>Section 3.3</u> and <u>Article IV</u>, the underwriting agreement into which each Holder and the Company shall enter shall contain such representations and warranties, covenants, indemnities and other rights and obligations of the Company and such Holders as are customary in underwritten offerings of securities. Under no circumstances shall the Company be obligated to effect (x) more than an aggregate of three (3) Underwritten Offerings pursuant to an Underwritten Demand by the Holders under this subsection 2.1.3 with respect to any or all Registrable Securities held by such Holders and (y) more than two (2) Underwritten Offerings per year pursuant to this subsection 2.1.3; provided, however, that an Underwritten Offering pursuant to an Underwritten Demand shall not be counted for such purposes unless a Registration Statement that may be available at such time has become effective and all of the Registrable Securities requested by the Requesting Holders and the Demanding Holders to be registered on behalf of the Requesting Holders and the Demanding Holders in such Registration Statement have been sold, in accordance with Section 3.1 of this Agreement.

2.1.4 Holder Information Required for Participation in Underwritten Offering. At least ten (10) business days prior to the first anticipated filing date of a Registration Statement pursuant to this Article II, the Company shall use reasonable best efforts to notify each Holder in writing (which may be by email) of the information reasonably necessary about the Holder to include such Holder's Registrable Securities in such Registration Statement. Notwithstanding anything else in this Agreement, the Company shall not be obligated to include such Holder's Registrable Securities to the extent the Company has not received such information, and received any other reasonably requested agreements or certificates, on or prior to the fifth business day prior to the first anticipated filing date of a Registration Statement pursuant to this Article II.

2.1.5 Reduction of Underwritten Offering. If the managing Underwriter or Underwriters in an Underwritten Offering, in good faith, advises the Company, the Demanding Holders and the Requesting Holders (if any) in writing that the dollar amount or number of Registrable Securities that the Demanding Holders and the Requesting Holders (if any) desire to sell, taken together with all other Common Stock or other equity securities that the Company desires to sell and the Common Stock, if any, as to which a Registration has been requested pursuant to separate written contractual piggyback registration rights held by any other stockholders who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders (pro rata based on the respective number of Registrable Securities that each Demanding Holder has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders have requested be included in such Underwritten Registration (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Registrable Securities of Requesting Holders (Pro Rata, based on the respective number of Registrable Securities that each Requesting Holder has so requested) exercising their rights to register their Registrable Securities pursuant to subsection 2.1.3 hereof, without exceeding the Maximum Number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), (ii) and (iii), the Common Stock or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities.

2.1.6 <u>Underwritten Offering Withdrawal</u>. A majority-in-interest of the Demanding Holders initiating an Underwritten Demand or a majority-in-interest of the Requesting Holders (if any), pursuant to a Registration under <u>subsection 2.1.3</u> shall have the right to withdraw from a Registration pursuant to an Underwritten Offering pursuant to <u>subsection 2.1.3</u> for any or no reason whatsoever upon written notification to the Company and the Underwriters (if any) of their intention to withdraw from such Registration at least five (5) business days prior to the effectiveness of the Registration Statement filed with the Commission with respect to the Registration of their Registrable Securities pursuant to such Underwritten Offering (or in the case of an Underwritten Registration pursuant to Rule 415, at least five (5) business days prior to the time of pricing of the applicable offering). Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to an Underwritten Offering prior to its withdrawal under this <u>subsection 2.1.6</u>.

2.2 Piggyback Registration.

2.2.1 Piggyback Rights. If, at any time on or after the Closing, the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company, including, without limitation, pursuant to Section 2.1 hereof), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for a rights offering or an exchange offer or offering of securities solely to the Company's existing stockholders, (iii) for an offering of debt that is convertible into equity securities of the Company or (iv) for a dividend reinvestment plan, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than ten (10) days before the anticipated filing date of such Registration Statement, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (such Registration a "Piggyback Registration"). The Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and shall use its reasonable best efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering to permit the Registrable Securities requested by the Holders pursuant to this subsection 2.2.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company included in such Registration and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this subsection 2.2.1 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.

- 2.2.2 <u>Reduction of Piggyback Registration</u>. If the managing Underwriter or Underwriters in an Underwritten Registration that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of the shares of Common Stock that the Company desires to sell, taken together with (i) the shares of Common Stock, if any, as to which Registration has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder, (ii) the Registrable Securities as to which registration has been requested pursuant to <u>Section 2.2</u> hereof, and (iii) the shares of Common Stock, if any, as to which Registration has been requested pursuant to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:
- (a) If the Registration is undertaken for the Company's account, the Company shall include in any such Registration (A) first, the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to <u>subsection 2.2.1</u> hereof, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the Common Stock, if any, as to which Registration has been requested or demanded pursuant to written contractual Piggyback Registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities;
- (b) If the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration (A) first, the Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A), (B) and (C), the Common Stock or other equity securities for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities.
- 2.2.3 <u>Piggyback Registration Withdrawal</u>. Any Holder of Registrable Securities shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration at least five (5) business days prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Piggyback Registration (or in the case of an Underwritten Registration pursuant to Rule 415, at least five (5) business days prior to the time of pricing of the applicable offering). The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this <u>subsection 2.2.3</u>.
- 2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to <u>Section 2.2</u> hereof shall not be counted as a Registration pursuant to an Underwritten Offering effected under <u>subsection 2.1.3</u>.
- 2.3 Restrictions on Registration Rights. Notwithstanding anything to the contrary contained herein, the Company shall not be obligated to (but may, at its sole option) (A) effect an Underwritten Offering (i) within sixty (60) days after the closing of an Underwritten Offering, or (ii) during the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, a Company-initiated Registration and provided that the Company has delivered written notice to the Holders prior to receipt of an Underwritten Demand pursuant to subsection 2.1.3 and it continues to actively employ, in good faith, all reasonable best efforts to cause the applicable Registration Statement to become effective or (B) file a Registration Statement (or any amendment thereto) or effect an Underwritten Offering (or, if the Company has filed a shelf Registration Statement and has included Registrable Securities therein, the Company shall be entitled to suspend the offer and sale of Registrable Securities pursuant to such Registration Statement) for a period of up to forty-five (45) days (i) if the Holders have requested an Underwritten Demand and the Company and the Holders are unable to obtain the commitment of Underwriters to firmly underwrite the offer; or (ii) in the good faith judgment of the Board such Underwritten Offering would be materially detrimental to the Company and the Board concludes as a result that it is essential to defer the filing of such Registration Statement at such time, provided that in each case of (i) and (ii) the Company shall furnish to such Holders a certificate signed by the Chairman of the Board stating that in the good faith judgment of the Board it would be materially detrimental to the Company for such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement and provided, further, that the Compa

ARTICLE III COMPANY PROCEDURES

- 3.1 <u>General Procedures</u>. If the Company is required to effect the Registration of Registrable Securities, subject to applicable law and any regulations promulgated by any securities exchange on which the Registrable Securities are then listed, the Company shall use its reasonable best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall, as expeditiously as possible:
- 3.1.1 prepare and file with the Commission within thirty (30) days a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold;
- 3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be reasonably requested by any Holder or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus;
- 3.1.3 prior to filing a Registration Statement or the Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and each Holder of Registrable Securities included in such Registration, and such Holder's legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and each Holder of Registrable Securities included in such Registration or the legal counsel for any such Holders may reasonably request in order to facilitate the disposition of the Registrable Securities owned by such Holders;
- 3.1.4 prior to any public offering of Registrable Securities, use its reasonable best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as any Holder of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business or as a dealer in securities in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;
- 3.1.5 cause all such Registrable Securities to be listed on each securities exchange or automated quotation system on which similar securities issued by the Company are then listed;
- 3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;
 - 3.1.7 provide a CUSIP number for all Registrable Securities, not later than the effective date of such Registration Statement;
- 3.1.8 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;
- 3.1.9 at least five (5) days prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus or any document that is to be incorporated by reference into such Registration Statement or Prospectus, furnish a copy thereof to each seller of such Registrable Securities and its counsel, including, without limitation, providing copies promptly upon receipt of any comment letters received with respect to any such Registration Statement or Prospectus;

- 3.1.10 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in Section 3.4 hereof;
- 3.1.11 permit a representative of the Holders (such representative to be selected by a majority of the participating Holders), the Underwriter(s), if any, and any attorney or accountant retained by such Holders or Underwriter(s) to participate, at each such person's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; provided, however, that any such representative or Underwriter enters into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information; and provided further, the Company may not include the name of any Holder or Underwriter or any information regarding any Holder or Underwriter in any Registration Statement or Prospectus, any amendment or supplement to such Registration Statement or Prospectus, any document that is to be incorporated by reference into such Registration Statement or Prospectus, or any response to any comment letter, without the prior written consent of such Holder or Underwriter and providing each such Holder or Underwriter a reasonable amount of time to review and comment on such applicable document, which comments the Company shall include unless contrary to applicable law;
- 3.1.12 obtain a "comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Registration which the participating Holders may rely on, in customary form and covering such matters of the type customarily covered by "comfort" letters as the managing Underwriter may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders;
- 3.1.13 on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion and negative assurance letter, dated as of such date, of counsel representing the Company for the purposes of such Registration, addressed to the Holders, the placement agent or sales agent, if any, and the Underwriter(s), if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Holders, placement agent, sales agent, or Underwriter(s) may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to a majority in interest of the participating Holders;
- 3.1.14 in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;
- 3.1.15 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission), which requirement will be deemed satisfied if the Company timely files Forms 10-Q and 10-K, as may be required to be filed under the Exchange Act and otherwise complies with Rule 158 under the Securities Act;

- 3.1.16 if the Registration involves the Registration of Registrable Securities involving gross proceeds in excess of \$50,000,000, use its reasonable best efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter(s) in any Underwritten Offering; and
- 3.1.17 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders, in connection with such Registration.
- 3.2 <u>Registration Expenses</u>. Except as otherwise provided herein, the Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "Registration Expenses," all reasonable fees and expenses of any legal counsel representing any Holder.
- 3.3 Requirements for Participation in Underwritten Offerings. No person or entity may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person or entity (i) agrees to sell such person's or entity's securities on the basis provided in any underwriting arrangements approved by the Company and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements. Notwithstanding anything in this Agreement to the contrary, if any Holder does not provide to the Company in writing information and affidavits as the Company reasonably requests for use in connection with any Registration Statement or Prospectus, the Company may exclude such Holder's Registrable Securities from the applicable Registration Statement or Prospectus if the Company determines, based on the advice of outside legal counsel, that such information is necessary to effect the registration and such Holder continues thereafter to withhold such information. For the avoidance of doubt, the exclusion of a Holder's Registrable Securities as a result of this Section 3.3 shall not affect the registration of the other Registrable Securities to be included in such Registration.
- 3.4 <u>Suspension of Sales</u>; <u>Adverse Disclosure</u>. Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until it has received copies of a supplemented or amended Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice), or until it is advised in writing by the Company that the use of the Prospectus may be resumed. If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would require the Company to make an Adverse Disclosure or would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time, but in no event more than forty-five (45) days, determined in good faith by the Company to be necessary for such purpose. In the event the Company exercises its rights under the preceding sentence, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities. The Company shall immediately notify the Holders of the expiration of any period during which it exercised its rights under this Section 3.4.

3.5 Reporting Obligations. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell shares of the Common Stock held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144, including providing any legal opinions. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

3.6 Lock-Up Restrictions.

3.6.1 During the Lock-Up Period, none of the Existing Holders shall offer, sell, contract to sell, pledge, grant any option to purchase, make any short sale or otherwise dispose of or distribute any shares of Common Stock that are subject to the Lock-Up Period or any securities convertible into, exercisable for, exchangeable for or that represent the right to receive shares of Common Stock that are subject to the Lock-Up Period, whether now owned or hereinafter acquired, that are owned directly by such Existing Holder (including securities held as a custodian) or with respect to which such Existing Holder has beneficial ownership within the rules and regulations of the Commission (such securities that are subject to the Lock-Up Period, the "Restricted Securities"), other than any transfer to an affiliate of an Existing Holder or to a Permitted Transferee, as applicable. The foregoing restriction is expressly agreed to preclude each Existing Holder, as applicable, from engaging in any hedging or other transaction with respect to Restricted Securities which is designed to or which reasonably could be expected to lead to or result in a sale or disposition of the Restricted Securities even if such Restricted Securities would be disposed of by someone other than such Existing Holder. Such prohibited hedging or other transactions include any short sale or any purchase, sale or grant of any right (including any put or call option) with respect to any of the Restricted Securities of the applicable Existing Holder, or with respect to any security that includes, relates to, or derives any significant part of its value from such Restricted Securities.

3.6.2 Each Existing Holder hereby represents and warrants that it now has and, except as contemplated by this <u>subsection 3.6.2</u> for the duration of the Lock-Up Period, will have good and marketable title to its Restricted Securities, free and clear of all liens, encumbrances, and claims that could impact the ability of such Existing Holder to comply with the foregoing restrictions Each Existing Holder agrees and consents to the entry of stop transfer instructions with the Company's transfer agent and registrar against the transfer of any Restricted Securities during the Lock-Up Period.

ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

4.1 Indemnification.

4.1.1 The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers and directors and each person who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and out-of-pocket expenses (including reasonable and documented attorneys' fees) caused by any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, except insofar as the same are caused by or contained in any information furnished in writing to the Company by such Holder expressly for use therein. The Company shall indemnify the Underwriters, their officers and directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.

- 4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by law, shall indemnify the Company, its directors and officers and agents and each person who controls the Company (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and out-of-pocket expenses (including without limitation reasonable attorneys' fees) resulting from any untrue statement of material fact contained in the Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information or affidavit so furnished in writing by such Holder expressly for use therein; provided, however, that the obligation to indemnify under this Section 4.1.2 shall be several, not joint and several, among the Holders of Registrable Securities, and the total liability of a Holder under this Section 4.1.2 shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement. The Holders of Registrable Securities shall indemnify the Underwriters, their officers, directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to indemnification of the Company.
- 4.1.3 Any person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided, that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party), and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel (plus local counsel) for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnified party, or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.
- 4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling person of such indemnified party and shall survive the transfer of securities.
- 4.1.5 If the indemnification provided under Section 4.1 hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; provided, however, that the liability of any Holder under this subsection 4.1.5 shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in subsections 4.1.1, 4.1.2 and 4.1.3 above, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this subsection 4.1.5 were determined by Pro Rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this subsection 4.1.5. No person guilty of fraudulent mi

ARTICLE V MISCELLANEOUS

5.1 Notices. Any notice or communication under this Agreement must be in writing and given by (i) deposit in the United States mail, addressed to the party to be notified, postage prepaid and registered or certified with return receipt requested, (ii) delivery in person or by courier service providing evidence of delivery, or (iii) transmission by hand delivery, facsimile or electronic mail. Each notice or communication that is mailed, delivered, or transmitted in the manner described above shall be deemed sufficiently given, served, sent, and received, in the case of mailed notices, on the third business day following the date on which it is mailed and, in the case of notices delivered by courier service, hand delivery, facsimile or electronic mail, at such time as it is delivered to the addressee (except in the case of electronic mail, with the delivery receipt or the affidavit of messenger) or at such time as delivery is refused by the addressee upon presentation. Any notice or communication under this Agreement must be addressed, if to the Company, to: [Banzai Holdings, Inc.], 435 Ericksen Ave, Suite 250, Bainbridge Island, WA 98110, Attn: Joseph Davy, E-mail: joe@banzai.io, with a copy to: Cooley LLP, 1700 Seventh Avenue, Suite 1900, Seattle, WA 98101-1355, Attn: Sonya Erickson, E-mail: serickson@cooley.com, and, if to any Holder, at such Holder's address or contact information as set forth in the Company's books and records. Any party may change its address for notice at any time and from time to time by written notice to the other parties hereto, and such change of address shall become effective thirty (30) days after delivery of such notice as provided in this Section 5.1.

5.2 Assignment; No Third Party Beneficiaries.

5.2.1 This Agreement and the rights, duties and obligations of the Company and the Holder of Registrable Securities, as the case may be, hereunder may not be assigned or delegated by the Company or the Holders of Registrable Securities, as the case may be, in whole or in part, except in connection with a transfer of Registrable Securities by such Holder to a Permitted Transferee but only if such Permitted Transferee agrees to become bound by the transfer restrictions set forth in this Agreement.

5.2.2 Prior to the expiration of the Lock-Up Period, no Holder subject to the Lock-Up Period may assign or delegate such Holder's rights, duties or obligations under this Agreement, in whole or in part, in violation of the Lock-Up Period, except in connection with a transfer of Registrable Securities by such Holder to a Permitted Transferee but only if such Permitted Transferee agrees to become bound by the transfer restrictions set forth in this Agreement.

- 5.2.3 This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors and the permitted assigns of the Holders, which shall include Permitted Transferees.
- 5.2.4 This Agreement shall not confer any rights or benefits on any persons that are not parties hereto, other than as expressly set forth in this Agreement and Section 5.2 hereof.
- 5.2.5 No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in <u>Section 5.1</u> hereof and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any transfer or assignment made other than as provided in this <u>Section 5.2</u> shall be null and void.
- 5.3 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts (including facsimile or PDF counterparts), each of which shall be deemed an original, and all of which together shall constitute the same instrument, but only one of which need be produced. Each party agrees that an electronic copy of this Agreement shall be considered and treated like an original, and that an electronic or digital signature shall be as valid as a handwritten signature (including .pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000 (e.g., www.docusign.com)).
- 5.4 <u>Governing Law; Venue.</u> NOTWITHSTANDING THE PLACE WHERE THIS AGREEMENT MAY BE EXECUTED BY ANY OF THE PARTIES HERETO, THE PARTIES EXPRESSLY AGREE THAT (I) THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF NEW YORK AS APPLIED TO AGREEMENTS AMONG NEW YORK RESIDENTS ENTERED INTO AND TO BE PERFORMED ENTIRELY WITHIN NEW YORK, WITHOUT REGARD TO THE CONFLICT OF LAW PROVISIONS OF SUCH JURISDICTION AND (II) THE VENUE FOR ANY ACTION TAKEN WITH RESPECT TO THIS AGREEMENT SHALL BE ANY STATE OR FEDERAL COURT IN NEW YORK COUNTY IN THE STATE OF NEW YORK.
- 5.5 Amendments and Modifications. Upon the written consent of the Company and the Holders of at least a majority in interest of the Registrable Securities at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; provided, however, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects either the Existing Holders as a group or the New Holders as a group, as the case may be, in a manner that is materially adversely different from the New Holders or the Existing Holders, respectively, shall require the consent of at least a majority-in-interest of the Registrable Securities held by such Existing Holders or a majority-in-interest of the Registrable Securities held by the New Holders, as applicable, at the time in question so affected; provided, further, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects one Holder or group of affiliated Holders, solely in its capacity as a holder of the shares of capital stock of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder or group of affiliated Holders so affected. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party. Any amendment, termination, or waiver effected in accordance with this Section 5.5 shall be binding on each party hereto and all of such party's successors and permitted assigns, regardless of whether or no

- 5.6 Other Registration Rights. The Company represents and warrants that no person, other than (i) a Holder of Registrable Securities and (ii) a holder of securities of the Company that are registrable pursuant to the GEM Agreement, has any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities for its own account or for the account of any other person. Further, the Company represents and warrants that, except with respect to the GEM Agreement, this Agreement supersedes the Existing Registration Rights Agreement and any other registration rights agreement or agreement with similar terms and conditions and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail. Notwithstanding the foregoing, the Company and the Holders hereby acknowledge that the Company has granted resale registration rights to certain holders of Company securities in the GEM Agreement, and that nothing herein shall restrict the ability of the Company to fulfill its resale registration obligations under the GEM Agreement.
- 5.7 Opt-Out Requests. Each Holder shall have the right, at any time and from time to time (including after receiving information regarding any potential public offering), to elect to not receive any notice that the Company or any other Holders otherwise are required to deliver pursuant to this Agreement by delivering to the Company a written statement signed by such Holder that it does not want to receive any notices hereunder (an "Opt-Out Request"); in which case and notwithstanding anything to the contrary in this Agreement the Company and other Holders shall not be required to, and shall not, deliver any notice or other information required to be provided to Holders hereunder to the extent that the Company or such other Holders reasonably expect would result in a Holder acquiring material non-public information within the meaning of Regulation FD promulgated under the Exchange Act. An Opt-Out Request may state a date on which it expires or, if no such date is specified, shall remain in effect indefinitely. A Holder who previously has given the Company an Opt-Out Request may revoke such request at any time, and there shall be no limit on the ability of a Holder to issue and revoke subsequent Opt-Out Requests; provided that each Holder shall use commercially reasonable efforts to minimize the administrative burden on the Company arising in connection with any such Opt-Out Requests.
- 5.8 Existing Registration Rights Agreement. The Sponsor and the Existing Holders hereby agree that upon execution of this Agreement by the Sponsor and the Existing Holders, the Existing Registration Rights Agreement shall be automatically terminated and superseded in its entirety by this Agreement.
- 5.9 Term. This Agreement shall terminate upon the earlier of (i) the seventh (7th) anniversary of the date of this Agreement, and (ii) the date as of which all of the Registrable Securities have been sold or disposed of or, (iii) with respect to any particular holder, (A) such Holder is permitted to sell the Registrable Securities held by him, her or it under Rule 144 (or any similar provision) under the Securities Act without limitation on the amount of securities sold or the manner of sale and without compliance with the current public reporting requirements set forth under Rule 144(i)(2), or (B) such Holder requests and the Company agrees to authorize the Company's transfer agent to remove any legend on share certificates of such Holder's Registrable Securities restricting further transfer (or any similar restriction in book entry positions of such Holder). The provisions of Section 3.5 and Article IV shall survive any termination.
- 5.8 Entire Agreement. This Agreement constitutes the entire understanding and agreement between the parties as to the matters covered herein and supersedes and replaces any prior understanding, agreement or statement of intent, in each case, written or oral, of any and every nature with respect thereto.

[Signature Page Follows]

	COMPANY:
	[BANZAI HOLDINGS INC.], a Delaware corporation
	By: Name:
	Title:
	SPONSOR:
	7GC & CO. HOLDINGS LLC, a Delaware limited liability company
	Ву:
	Name: Title:
	OTHER PRE-IPO HOLDERS:
	Courtney Robinson
	Tripp Jones
	Kent Schofield
	Patrick Eggen
	NEW HOLDERS:
	[•]
[Signatu	re Page to Amended and Restated Registration Rights Agreement]
	B-18

AMENDED AND RESTATED COMPANY SUPPORT AGREEMENT

This AMENDED AND RESTATED COMPANY SUPPORT AGREEMENT (this "<u>Agreement</u>") is dated as of August 4, 2023, by and among 7GC & Co. Holdings Inc., a Delaware corporation ("<u>7GC</u>"), the Persons set forth on <u>Schedule I</u> hereto (each, a "<u>Stockholder</u>" and, collectively, the "<u>Stockholders</u>"), and Banzai International, Inc., a Delaware corporation (the "<u>Company</u>"). Capitalized terms used but not defined herein shall have the respective meanings ascribed to such terms in the Merger Agreement (as defined below).

RECITALS

WHEREAS, on December 8, 2022, 7GC, the Company and the Stockholders entered into that certain Company Support Agreement (the "Original Company Support Agreement");

WHEREAS, concurrently with the execution of the Original Company Support Agreement, 7GC, 7GC Merger Sub I , Inc., a Delaware corporation and an indirect wholly owned subsidiary of 7GC ("First Merger Sub"), 7GC Merger Sub II, LLC, a Delaware limited liability company and a direct wholly owned subsidiary of 7GC ("Second Merger Sub" and, together with First Merger Sub, the "Merger Subs" and each, a "Merger Sub"), and the Company entered into an Agreement and Plan of Merger and Reorganization (the "Original Merger Agreement"), pursuant to which, among other transactions, upon the terms and subject to the conditions thereof, at the Closing, First Merger Sub will merge with and into the Company (the "First Merger"), with the separate corporate existence of First Merger Sub ceasing and the Company surviving the First Merger as an indirect wholly owned subsidiary of 7GC (the "Surviving Corporation") and, promptly following the First Merger, but in any event on the same day as the First Merger and as part of the same overall transaction as the First Merger, the Surviving Corporation will merge with and into Second Merger Sub (the "Second Merger" and, together with the First Merger, the "Mergers"), with Second Merger Sub surviving the Second Merger as a wholly owned subsidiary of 7GC, and, upon the First Effective Time of the First Merger, each share of Company Stock issued and outstanding as of immediately prior to the First Effective Time shall be cancelled and automatically converted into the right to receive the consideration as described in the Original Merger Agreement (such transaction, the Mergers and the other transactions contemplated by the Original Merger Agreement, the "Transactions"), and a condition to the closing of the Transactions pursuant to the Original Merger Agreement is the completion of the Company's acquisition of Hyros Inc., a Delaware corporation ("Hyros");

WHEREAS, 7GC and the Company have decided to pursue the closing of the Transactions without completing the Company's acquisition of Hyros and requiring the completion of such acquisition as a condition to the closing of the Transactions and, concurrently with the execution of this Agreement, the Company, 7GC, and the Merger Subs are entering into that certain Amendment to Agreement Plan of Merger and Reorganization (the "Merger Agreement Amendment," and the Original Merger Agreement as amended by the Merger Agreement Amendment, the "Merger Agreement") to reflect this structural change and the corresponding changes to certain terms of the Original Merger Agreement;

WHEREAS, as of the date hereof, the Stockholders are the holders of record and "beneficial owners" (within the meaning of Rule 13d-3 of the Exchange Act) of such numbers and classes of shares of Company Stock as are indicated opposite their names on <u>Schedule I</u> attached hereto (all such Company Stock, together with any New Securities (as defined below), are referred to herein as the "<u>Subject Shares</u>");

WHEREAS, the Company and the Stockholders hereby agree to terminate, effective as of the Closing, each Company Stockholder Agreement; and

WHEREAS, pursuant to Section 2 of the Merger Agreement Amendment and as an inducement to 7GC entering into the Merger Agreement Amendment and to consummate the Transactions, 7GC, the Company and the Stockholders desire to (i) amend and restate, in its entirety, the Original Company Support Agreement by entering into this Agreement to support the closing of the Transactions pursuant to the terms of the Merger Agreement and (ii) agree to the matters as set forth herein.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing and of the mutual covenants and agreements contained herein, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

ARTICLE 1

COMPANY SUPPORT AGREEMENT; COVENANTS

1.1 Transfer of Shares. During the period commencing on the date hereof and ending on the earlier to occur of (a) the Closing, and (b) such date and time as the Merger Agreement shall be terminated in accordance with Section 7.1 of the Merger Agreement (the "Expiration Time"), each Stockholder agrees that it shall not (i) sell, assign, offer, exchange, transfer (including by operation of law), pledge, dispose of, permit to exist any Lien with respect to, or otherwise encumber (each, a "Transfer"), any of the Subject Shares or otherwise agree or commit to do any of the foregoing, (ii) deposit any Subject Shares into a voting trust or enter into a voting agreement or arrangement or grant any proxy or power of attorney with respect thereto that is inconsistent with this Agreement, (iii) enter into any contract, option or other arrangement or undertaking with respect to the direct or indirect acquisition or sale, assignment, transfer (including by operation of law) or other disposition of any Subject Shares or (iv) take any action that would have the effect of preventing, impeding, interfering with or adversely affecting such Stockholder's ability to perform its obligations under this Agreement (including pursuant to Section 1.3); provided, that the foregoing shall not prohibit the transfer of the Subject Shares, (A) if Stockholder is an individual (1) to any member of such Stockholder's immediate family, or to a trust for the benefit of Stockholder or any member of Stockholder's immediate family, the sole trustees of which are such Stockholder or any member of such Stockholder's immediate family, (2) by will, other testamentary document, under the Laws of intestacy or by virtue of Laws of descent and distribution upon the death of Stockholder, or (3) by operation of law pursuant to a qualified domestic relations order or in connection with a divorce settlement; (B) if Stockholder is an entity, to a partner, member, or Affiliate of Stockholder; or (C) if Stockholder is a trust, any beneficiary of such Stockholder or the estate of any such beneficiary; but only if, in the case of clauses (A), (B) and (C), such transferee shall concurrently execute this Agreement or a joinder agreeing to become a party to this Agreement. Any attempted transfer of Subject Shares or any interest therein in violation of this <u>Section 1.1</u> shall be null and void.

- 1.2 New Shares. In the event that, during the period commencing on the date hereof and ending at the Expiration Time, (a) any shares of Company Stock are issued to a Stockholder after the date of this Agreement pursuant to any stock dividend, stock split, recapitalization, reclassification, combination or exchange of Subject Shares or otherwise, (b) a Stockholder purchases or otherwise acquires beneficial ownership of any shares of Company Stock or (c) a Stockholder acquires the right to vote or share in the voting of any shares of Company Stock (including, without limitation, by proxy or power of attorney) (collectively the "New Securities"), then such New Securities acquired or purchased by such Stockholder shall be subject to the terms of this Agreement to the same extent as if they constituted the Subject Shares owned by such Stockholder as of the date hereof. Each of the Stockholders agrees, while this Agreement is in effect, to notify the Company promptly pursuant to the provisions of Section 3.8 of the number of New Securities acquired by such Stockholder, if any, after the date hereof.
- 1.3 <u>Agreement to Vote</u>. During the period commencing on the date hereof until the Expiration Time, each Stockholder, with respect to its, his or her Subject Shares unconditionally and irrevocably agrees that, at any meeting of the stockholders of the Company requested by the Board of Directors of the Company or undertaken as contemplated by the Transactions (or any adjournment or postponement thereof), and in any action by written consent of the stockholders of the Company (which written consent shall be delivered promptly, and in any event within twenty four (24) hours, after (x) the Registration Statement / Proxy Statement has been declared effective and (y) the Company requests such delivery), such Stockholder shall, if a meeting is held, appear at the meeting, in person or by proxy, or otherwise cause its, his or her Subject Shares to be counted as present thereat for purposes of establishing a quorum, and such Stockholder shall vote or provide consent (or cause to be voted or consented), in person or by proxy, all of its, his or her Subject Shares:
 - (a) to approve and adopt the Merger Agreement and the Transactions (including, but not limited to, the Company Preferred Conversion);
- (b) in any other circumstances upon which a consent or other approval is required under the Governing Documents of the Company or the Company Stockholder Agreements or otherwise sought with respect to, or in connection with, the Merger Agreement or the Transactions, to vote, consent or approve (or cause to be voted, consented or approved) with respect to all of such Stockholder's Subject Shares held at such time in favor thereof; and
- (c) against any 7GC Competing Transaction or any proposal, action or agreement that would impede, interfere, frustrate, delay, postpone, prevent or nullify any provision of this Agreement, the Merger Agreement or the Merger.

Each Stockholder hereby agrees that it shall not commit or agree to take any action inconsistent with the foregoing. Notwithstanding the foregoing, to the extent such Stockholder is not a director, officer or Affiliate of the Company or holder of Subject Shares representing greater than 5% of the outstanding shares of capital stock of the Company, such Stockholder shall not be obligated to vote or provide consent with respect to any of its, his or her Subject Shares or take any other action, in each case solely to the extent any such vote, consent or other action would preclude 7GC from filing with the SEC the Registration Statement / Proxy Statement on Form S-4 as contemplated by the Merger Agreement.

- 1.4 No Adverse Action; Stop Order. Each of the Stockholders agrees, while this Agreement is in effect, not to take or agree or commit to take any action that would make any representation and warranty of such Stockholder contained in this Agreement inaccurate in any material respect. Each of the Stockholders further agrees that it shall use its commercially reasonable efforts to cooperate with 7GC to effect the transactions contemplated hereby and the Transactions. In furtherance of the foregoing, the Company hereby agrees (a) to place a revocable stop order on all Subject Shares and (b) not to process any attempts by the Stockholders to Transfer any Subject Shares except in compliance with the terms of this Agreement.
- 1.5 No Challenges. Each Stockholder agrees not to commence, join in, facilitate, assist or encourage, and agrees to take all actions necessary to opt out of any class in any class action with respect to, any claim, derivative or otherwise, against 7GC, First Merger Sub, Second Merger Sub, the Company or any of their respective successors or directors (a) challenging the validity of, or seeking to enjoin the operation of, any provision of this Agreement, (b) alleging a breach of any fiduciary duty of any person in connection with the evaluation, negotiation or entry into the Merger Agreement or (c) otherwise relating to the negotiation, execution or delivery of this Agreement, the Merger Agreement or the consummation of the transactions contemplated hereby and thereby.
- 1.6 <u>Termination of Company Stockholder Agreements</u>. Each of the Company and each Stockholder hereby agrees and consents to the termination of all Company Stockholder Agreements to which such Stockholder is party, effective as of the Closing, without any further liability or obligation to such Stockholder, the Company, the Company's Subsidiaries, or 7GC. Each Stockholder agrees, confirms, and acknowledges that, after the Closing, it, he or she shall not have any of the rights or privileges provided to each such Stockholder in such applicable Company Stockholder Agreements. The termination of such Company Stockholder Agreements shall terminate the rights of the parties thereto to enforce any provisions of such agreements that expressly survive the termination of such Company Stockholder Agreements.
- 1.7 <u>Registration Rights Agreement</u>. Each of the Stockholders set forth on <u>Schedule II</u> (and any Person to whom each such Stockholder transfers its Subject Shares as permitted by the terms of this Agreement) will deliver, at the Closing, a duly executed copy of the Registration Rights Agreement, substantially in the form attached as Exhibit B to the Merger Agreement.
- 1.8 <u>Lock-up Agreements</u>. Each of the Stockholders set forth on <u>Schedule III</u> (and any Person to whom each such Stockholder transfers its Subject Shares as permitted by the terms of this Agreement) will deliver, at the Closing, a duly executed copy of the Lock-up Agreement, substantially in the form attached as Exhibit C to the Merger Agreement.
- 1.9 <u>Appraisal and Dissenters' Rights</u>. Each Stockholder hereby irrevocably and unconditionally waives, and agrees not to assert or perfect, any rights of appraisal or other similar rights to dissent (including any notice requirements related thereto) with respect to the Merger, the Merger Agreement or any of the Transactions that Stockholder may have by virtue of ownership of Subject Shares (including all rights under Section 262 of the DGCL).

1.10 Exclusivity. Unless this Agreement shall have been terminated in accordance with Section 3.1, each Stockholder agrees not to, and shall cause its, his or her Affiliates or Representatives not to, (a) accept, initiate, respond to, encourage, entertain, solicit, negotiate, provide information with respect to or discuss other offers for an Acquisition Proposal; (b) furnish or disclose any non-public information to any Person in connection with, or that could reasonably be expected to lead to, an Acquisition Proposal; (c) enter into any Contract regarding an Acquisition Proposal; (d) prepare or take any steps in connection with a public offering of any Equity Securities of any Group Company (or any successor to or parent company of any Group Company); or (e) otherwise cooperate in any way with, or assist or participate in, or facilitate or encourage any effort or attempt by any Person to do or seek to do any of the foregoing or seek to circumvent this Section 1.10 or further an Acquisition Proposal. Each Stockholder agrees to (A) notify the Company and 7GC promptly upon receipt (and in any event within forty-eight (48) hours after receipt) of any Acquisition Proposal, and to describe the terms and conditions of any such Acquisition Proposal in reasonable detail (including the identity of the Persons making such Acquisition Proposal), (B) keep the Company and 7GC fully informed on a current basis of any material modifications to such offer or information and (C) not (and shall cause its Affiliates and Representatives not to) conduct any further discussions with, provide any information to, or enter into negotiations with such Persons. Each Stockholder shall immediately cease and cause to be terminated any discussions or negotiations with any Persons (other than 7GC and its Representatives) that may be ongoing with respect to an Acquisition Proposal and terminate any such Person's and such Person's Representative's access to any electronic data room. Each Stockholder shall not release any third party from, or wa

1.11 Consent to Disclosure. Each Stockholder hereby consents to the publication and disclosure in the Registration Statement / Proxy Statement (and, as and to the extent otherwise required by applicable securities Laws or the SEC or any other securities authorities, any other documents or communications provided by the Company or 7GC to any Governmental Entity or to securityholders of 7GC) of such Stockholder's identity and beneficial ownership of its Subject Shares and the nature of such Stockholder's commitments, arrangements and understandings under and relating to this Agreement and, if deemed appropriate by the Company or 7GC, a copy of this Agreement. Each Stockholder will promptly provide any information reasonably requested by the Company or 7GC in connection with the first sentence of this Section 1.11 or for any regulatory application or filing made or approval sought in connection with the Transactions (including filings with the SEC).

REPRESENTATIONS AND WARRANTIES

2.1 <u>Representations and Warranties of the Stockholders</u>. Each Stockholder represents and warrants as of the date hereof to 7GC and the Company (solely with respect to itself, himself or herself and not with respect to any other Stockholder) as follows:

(a) <u>Organization</u>; <u>Due Authorization</u>. If such Stockholder is not an individual, it is duly organized, validly existing and in good standing under the Laws of the jurisdiction in which it is incorporated, formed, organized or constituted, and the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby are within such Stockholder's corporate, limited liability company or organizational powers and have been duly authorized by all necessary corporate, limited liability company or organizational actions on the part of such Stockholder. If such Stockholder is an individual, such Stockholder has full legal capacity, right and authority to execute and deliver this Agreement and to perform his or her obligations hereunder. This Agreement has been duly executed and delivered by such Stockholder and, assuming due authorization, execution and delivery by the other parties to this Agreement, this Agreement constitutes a legally valid and binding obligation of such Stockholder, enforceable against such Stockholder in accordance with the terms hereof (except as enforceability may be limited by bankruptcy Laws, other similar Laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies). If any Stockholder is an individual and is married, and any of the Subject Shares of such Stockholder constitute community property or spousal approval is otherwise necessary for this Agreement to be legal, binding and enforceable with respect to such Stockholder, such Stockholder's spouse has delivered with this Agreement a Spousal Consent in the form attached hereto as <u>Exhibit A</u> and this Agreement has been duly authorized, executed and delivered by, and constitutes the legal, valid and binding obligation of, such Stockholder's spouse, enforceable against such Stockholder's spouse in accordance with its terms. If this Agreement is being executed in a representative or fiduciary capacity, the Perso

(b) Ownership. Such Stockholder is the record and beneficial owner (as defined in the Securities Act) of, and has good, valid and marketable title to, all of such Stockholder's Subject Shares, and there exist no Liens or any other limitation or restriction (including any restriction on the right to vote, sell or otherwise dispose of such Subject Shares) affecting any such Subject Shares, other than Liens pursuant to (i) this Agreement, (ii) the Governing Documents of the Company, (iii) the Company Stockholder Agreements, (iv) the Merger Agreement, or (v) any applicable securities Laws. Such Stockholder's Subject Shares are the only Equity Securities in the Company owned of record or beneficially by such Stockholder on the date of this Agreement, and none of such Stockholder's Subject Shares are subject to any proxy, voting trust or other agreement or arrangement with respect to the voting of such Subject Shares, except as provided hereunder and under the Company Stockholder Agreements. Such Stockholder has full and sole voting power (including the right to control such vote as contemplated herein), full power of disposition and full power to agree to all of the matters applicable to such Stockholder set forth in this Agreement, in each case, with respect to such Stockholder's Subject Shares (including, without limitation, by proxy or power of attorney). Such Stockholder does not hold or own any rights to acquire (directly or indirectly) any Equity Securities of the Company or any Equity Securities convertible into, or which can be exchanged for, Equity Securities of the Company. Such Stockholder has not granted a proxy or power of attorney with respect to any of the Stockholder's Subject Shares that is inconsistent with the Stockholder's obligations pursuant to this Agreement. There are no claims for finder's fees or brokerage commissions or other like payments in connection with this Agreement or the transactions contemplated hereby payable by such Stockholder pursuant to arrangements made by such Stockholder.

- (c) No Conflicts. The execution and delivery of this Agreement by such Stockholder does not, and the performance by such Stockholder of his, her or its obligations hereunder and the compliance by such Stockholder of his, her or its obligations hereunder do not and will not, (i) if such Stockholder is not an individual, contravene, conflict with or result in a violation of the organizational documents of such Stockholder or (ii) require any consent, authorization or approval that has not been given or other action that has not been taken by any Person (including under any Contract binding upon such Stockholder or such Stockholder's Subject Shares), in each case, to the extent such consent, approval or other action would prevent, enjoin or materially delay the performance by such Stockholder of its, his or her obligations under this Agreement, (iii) conflict with or violate any Law, or (iv) result in the creation of a Lien on any of the Subject Shares. No consent, approval, order or authorization of, or registration, declaration or filing with, any Governmental Entity or any other person is required by or with respect to such Stockholder in connection with the execution and delivery of this Agreement or the consummation by such Stockholder of the transactions contemplated hereby. If such Stockholder is an individual, no consent of such Stockholder's spouse is necessary under any "community property" or other Laws in order for such Stockholder to enter into and perform its obligations under this Agreement.
- (d) <u>Litigation</u>. There are no Proceedings pending against such Stockholder, or to the knowledge of such Stockholder threatened against such Stockholder, before (or, in the case of threatened Proceedings, that would be before) any arbitrator or any Governmental Entity, which in any manner challenges, seeks to prevent, enjoin or materially delay or could have the effect of preventing, enjoining or materially delaying the performance by such Stockholder of its, his or her obligations under this Agreement.
- (e) <u>Adequate Information</u>. Such Stockholder is a sophisticated stockholder and has adequate information concerning the business and financial condition of 7GC and the Company to make an informed decision regarding this Agreement, the Transactions and the transactions contemplated hereby and has independently and without reliance upon 7GC or the Company and based on such information as such Stockholder has deemed appropriate, made its, his or her own analysis and decision to enter into this Agreement. Such Stockholder acknowledges that 7GC and the Company have not made and do not make any representation or warranty, whether express or implied, of any kind or character except as expressly set forth in this Agreement. Such Stockholder acknowledges that the agreements contained herein with respect to the Subject Shares held by such Stockholder are irrevocable.
- (f) <u>Acknowledgment</u>. Such Stockholder understands and acknowledges that each of 7GC and the Company is entering into the Merger Agreement in reliance upon such Stockholder's execution and delivery of this Agreement.
- (g) No Inconsistent Agreements. Such Stockholder hereby covenants and agrees that, except for this Agreement, such Stockholder (a) has not entered into, nor will enter into at any time while this Agreement remains in effect, any voting agreement or voting trust with respect to such Stockholder's Subject Shares inconsistent with such Stockholder's obligations pursuant to this Agreement, (b) has not granted, nor will grant at any time while this Agreement remains in effect, a proxy, consent or power of attorney with respect to such Stockholder's Subject Shares and (c) has not entered into any agreement or knowingly taken any action (nor will enter into any agreement or knowingly take any action) that would make any representation or warranty of such Stockholder contained herein untrue or incorrect in any material respect or have the effect of preventing such Stockholder from performing any of its material obligations under this Agreement.

ARTICLE 3

MISCELLANEOUS

- 3.1 <u>Termination</u>. This Agreement and all of its provisions shall automatically terminate upon the earliest of (a) the Expiration Time and (b) as to each Stockholder, the written agreement of 7GC, the Company and such Stockholder. Upon such termination of this Agreement, all obligations of the parties under this Agreement will terminate, without any liability or other obligation on the part of any party hereto to any Person in respect hereof or the transactions contemplated hereby, and no party hereto shall have any claim against another (and no Person shall have any rights against such party), whether under contract, tort or otherwise, with respect to the subject matter hereof; <u>provided</u>, <u>however</u>, that the termination of this Agreement shall not relieve any party hereto from liability arising in respect of any breach of this Agreement prior to such termination. This <u>Article III</u> shall survive the termination of this Agreement.
- 3.2 <u>Governing Law</u>. This Agreement and all related Proceedings shall be governed by and construed in accordance with the internal Laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Law of any jurisdiction other than the State of Delaware.

3.3 CONSENT TO JURISDICTION AND SERVICE OF PROCESS; WAIVER OF JURY TRIAL.

(a) ANY PROCEEDING BASED UPON, ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY MUST BE BROUGHT IN THE COURT OF CHANCERY OF THE STATE OF DELAWARE (OR, TO THE EXTENT SUCH COURT DOES NOT HAVE SUBJECT MATTER JURISDICTION, THE SUPERIOR COURT OF THE STATE OF DELAWARE), OR, IF IT HAS OR CAN ACQUIRE JURISDICTION, IN THE UNITED STATES DISTRICT COURT FOR THE DISTRICT OF DELAWARE, AND EACH OF THE PARTIES IRREVOCABLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF EACH SUCH COURT IN ANY SUCH PROCEEDING, WAIVES ANY OBJECTION IT MAY NOW OR HEREAFTER HAVE TO PERSONAL JURISDICTION, VENUE OR TO CONVENIENCE OF FORUM, AGREES THAT ALL CLAIMS IN RESPECT OF THE PROCEEDING SHALL BE HEARD AND DETERMINED ONLY IN ANY SUCH COURT, AND AGREES NOT TO BRING ANY PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY IN ANY OTHER COURT. NOTHING HEREIN CONTAINED SHALL BE DEEMED TO AFFECT THE RIGHT OF ANY PARTY TO SERVE PROCESS IN ANY MANNER PERMITTED BY LAW OR TO COMMENCE LEGAL PROCEEDINGS OR OTHERWISE PROCEED AGAINST ANY OTHER PARTY IN ANY OTHER JURISDICTION, IN EACH CASE, TO ENFORCE JUDGMENTS OBTAINED IN ANY PROCEEDING BROUGHT PURSUANT TO THIS <u>SECTION 3.3(a)</u>.

(b) WAIVER OF TRIAL BY JURY. THE PARTIES EACH HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION, OR CAUSE OF ACTION (I) ARISING UNDER THIS AGREEMENT OR (II) IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES IN RESPECT OF THIS AGREEMENT OR ANY OF THE TRANSACTIONS RELATED HERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER IN CONTRACT, TORT, EQUITY, OR OTHERWISE. THE PARTIES EACH HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION, OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY AND THAT THE PARTIES MAY FILE AN ORIGINAL COUNTERPART OF A COPY OF THIS AGREEMENT WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE PARTIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

- 3.4 <u>Assignment</u>. This Agreement may not be assigned by any party (whether by operation of Law or otherwise) without the prior written consent of the other parties hereto. Any attempted assignment of this Agreement not in accordance with the terms of this <u>Section 3.4</u> shall be void, *ab initio*.
- 3.5 <u>Specific Performance</u>. The parties agree that irreparable damage for which monetary damages, even if available, would not be an adequate remedy, would occur in the event that the parties do not perform their respective obligations under the provisions of this Agreement (including failing to take such actions as are required of them hereunder to consummate the Transactions) in accordance with their specific terms or otherwise breach such provisions. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions, specific performance and other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, including the Stockholder's obligations to vote its Subject Shares as provided in this Agreement, in each case without posting a bond or undertaking and without proof of damages and this being in addition to any other remedy to which they are entitled at Law or in equity. Each of the parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief when expressly available pursuant to the terms of this Agreement on the basis that the other parties have an adequate remedy at Law or an award of specific performance is not an appropriate remedy for any reason at Law or equity.
- 3.6 <u>Amendment; Waiver</u>. This Agreement may be amended or modified only by a written agreement executed and delivered by 7GC, the Company and the Stockholders. This Agreement may not be modified or amended except as provided in the immediately preceding sentence and any purported amendment by any party or parties effected in a manner which does not comply with this <u>Section 3.6</u> shall be void, *ab initio*.
- 3.7 Severability. Whenever possible, each provision of this Agreement will be interpreted in such a manner as to be effective and valid under applicable Law, but if any term or other provision of this Agreement is held to be invalid, illegal or unenforceable under applicable Law, all other provisions of this Agreement shall remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision of this Agreement is invalid, illegal or unenforceable under applicable Law, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby are consummated as originally contemplated to the greatest extent possible.

3.8 <u>Notices</u>. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given) when delivered in person, when delivered by e-mail (having obtained electronic delivery confirmation thereof), or when sent by registered or certified mail (postage prepaid, return receipt requested) (upon receipt thereof) to the other parties as follows:

If to 7GC:

c/o 7GC & Co. Holdings LLC 388 Market Street, Suite 1300 San Francisco, CA 94111 Attention:Jack Leeney Chris Walsh

E-mail: jack@7GC.co chris@7GC.co

with a copy (which shall not constitute notice) to:

Sidley Austin LLP 1999 Avenue of the Stars 17th Floor Los Angeles, CA 90067 Attention:Joshua G. DuClos E-mail: jduclos@sidley.com

Sidley Austin LLP
One South Dearborn Street
Chicago, IL 60603
Attention:Michael P. Heinz and Matthew D. Stoker
E-mail: mheinz@sidley.com and mstoker@sidley.com

If to the Company:

Banzai International, Inc. 435 Ericksen Ave, Suite 250 Bainbridge Island, WA 98110 Attention:Joseph Davy E-mail: joe@banzai.io

with a copy (which shall not constitute notice) to:

Cooley LLP 1700 Seventh Avenue, Suite 1900 Seattle, WA 98101-1355 Attention:Sonya Erickson E-mail: serickson@cooley.com

If to a Stockholder:

To such Stockholder's address set forth on Stockholder's signature page hereto.

or to such other address as the party to whom notice is given may have previously furnished to the others in writing in the manner set forth above.

- 3.9 <u>Further Assurances</u>. Each Stockholder shall execute and deliver, or cause to be delivered, such additional documents, and take, or cause to be taken, all such further actions and do, or cause to be done, all things reasonably necessary (including under applicable Laws), or reasonably requested by 7GC or the Company, to effect the actions and consummate the Merger and the other transactions contemplated by this Agreement and the Merger Agreement (including the Transactions), in each case, on the terms and subject to the conditions set forth therein and herein, as applicable.
- 3.10 Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed to be an original, but all of which shall constitute one and the same agreement. The words "execution," "signed," "signature," and words of like import in this Agreement shall include images of manually executed signatures transmitted by facsimile or other electronic format (including, "pdf", "tif" or "jpg") and other electronic signatures (including, DocuSign and AdobeSign). The use of electronic signatures and electronic records (including, any contract or other record created, generated, sent, communicated, received, or stored by electronic means) shall be of the same legal effect, validity and enforceability as a manually executed signature or use of a paper-based record-keeping system to the fullest extent permitted by applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the Delaware Uniform Electronic Transactions Act and any other applicable law. Minor variations in the form of the signature page, including footers from earlier versions of this Agreement, shall be disregarded in determining the party's intent or the effectiveness of such signature.
- 3.11 Entire Agreement; Effect of this Agreement. This Agreement constitutes the full and entire agreement among the parties hereto with respect to the subject matter hereof and supersedes all other prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof. For the avoidance of doubt and in furtherance of the foregoing, upon the execution and delivery of this Agreement, by the Company, Stockholders and 7GC, the Original Support Agreement shall automatically terminate and be of no further force and effect and the Original Company Support Agreement is hereby amended and restated in its entirety as set forth in this Agreement.

[Remainder of Page Intentionally Left Blank]

7GC & CO. HOLDINGS INC.

By: /s/ Jack Leeney

Name: Jack Leeney

Title: Chairman and Chief Executive Officer

BANZAI INTERNATIONAL, INC.

By: /s/ Joseph P. Davy

Name: Joseph P. Davy
Title: Chief Executive Officer

/s/ Sheila Linteau

Sheila Linteau, as Administrator of the estate of Roland A. Linteau III, Deceased

Address: ****
Email: ****

/s/ Joseph P. Davy

Joseph P. Davy

Address: ****
Email: ****

DNX PARTNERS III, LP

By: DNX Venture Partners III, LP,

its General Partner

By: DNX III, LLC, its General Partner

By: /s/ Mitch Kitamura
Name: Mitch Kitamura
Title: Authorized Signatory

DNX PARTNERS JAPAN III, LP

By: DNX Venture Partners III, LP,

its General Partner

By: DNX III, LLC, its General Partner

By: /s/ Mitch Kitamura
Name: Mitch Kitamura
Title: Authorized Signatory

DNX PARTNERS S-III, LP

By: DNX Partners S3, LLC

its General Partner

By: /s/ Mitch Kitamura
Name: Mitch Kitamura
Title: Authorized Signatory

SCHEDULE I

Stockholder Subject Shares

Stockholder	Class of Shares	Number of Shares
DNX Partners III, LP	Series A-1 Preferred Stock	1,014,166
DNX Partners Japan III, LP	Series A-1 Preferred Stock	350,266
DNX Partners S-III, LP	Series A-1 Preferred Stock	7,545
Joseph P. Davy	Class B Common Stock	3,760,000
The Estate of Roland A. Linteau III	Class B Common Stock	2,560,000

SCHEDULE II

Registration Rights Agreement Signatories

- 1. Joseph Davy
- 2. Sheila Linteau on behalf of the estate of Roland Linteau
- 3. DNX Partners III, LP
- 4. DNX Partners Japan III, LP
- 5. DNX Partners S-III, LP

SCHEDULE III

Lock-up Agreement Signatories

- 1. Joseph Davy
- 2. Sheila Linteau on behalf of the estate of Roland Linteau
- 3. DNX Partners III, LP
- 4. DNX Partners Japan III, LP
- 5. DNX Partners S-III, LP

EXHIBIT A

Form of Spousal Consent

I,	, spouse of	("Stockholder"), acknowledge that I have read the Amended and Restated Company
Support Agreement,	executed by Stockholder with 7GC & C	Co. Holdings Inc., a Delaware corporation (" <u>7GC</u> "), and Banzai International, Inc., a Delaware
		nt"), and that I know the contents of the Agreement. Capitalized terms used but not defined
herein shall have the	meanings assigned to them in the Agree	ement.
including any interes irrevocably subject to	t that I might have therein. I understan o the Agreement and the other agreemen	s regarding Company Stock (as defined in the Merger Agreement) that my spouse may own, and and agree that my interest, if any, in any Company Stock subject to the Agreement shall be not referred to therein. I further understand and agree that any community property interest that to the Agreement and the other agreements referred to therein.
acknowledge, deliver necessary, incidental Company Stock or N without further signa of any and all such C that the foregoing po	r, file and record all documents which, convenient or otherwise), any and all lew Securities (as defined in the Agree ture, consent or knowledge, to exercise company Stock and New Securities), wiwer of attorney and proxy is coupled wi	is my true and lawful attorney and proxy in my name, place and stead to sign, make, execute, may be required, and to manage, vote, act and make all decisions with respect to (whether all Company Stock in which I now have or hereafter acquire any interest and in any and all ement) now or hereafter held of record by Stockholder (including but not limited to, the right, a amendments and modifications of, and to terminate, the foregoing agreements and to dispose ith all powers I would possess if personally present, it being expressly understood and intended ith an interest; and this power of attorney is a durable power of attorney and will not be affected ion of marriage and this proxy will not terminate without the consent of Stockholder, 7GC and
	with respect to this consent. I have eithe	ers contained in the Agreement are complex and that I am free to seek independent professional er sought such guidance or counsel or determined after carefully reviewing the Agreement that I
		
		Name:
		Date:

AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

This AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT (this "Agreement"), dated as of [●], 2023, is made and entered into by and among [Banzai Holdings, Inc.] (formerly known as 7GC & Co. Holdings Inc.), a Delaware corporation (the "Company"), 7GC & Co. Holdings LLC, a Delaware limited liability company (the "Sponsor"), each of the undersigned parties that holds Founder Shares (as defined below) and is identified as an "Other Pre-IPO Holder" on the signature pages hereto (collectively, with the Sponsor, the "Existing Holders"), and the undersigned parties identified as "New Holders" on the signature pages hereto (collectively, the "New Holders") (each of the foregoing parties (other than the Company) and any person or entity who hereafter becomes a party to this Agreement pursuant to Section 5.2 of this Agreement, a "Holder" and collectively, the "Holders"). Capitalized terms used but not otherwise defined in this Agreement shall have the meanings ascribed thereto in the Merger Agreement (as defined below).

RECITALS

WHEREAS, the Company and the Sponsor entered into that certain Securities Subscription Agreement, dated as of September 18, 2020, pursuant to which the Sponsor purchased an aggregate of 5,031,250 shares of the Company's Class B common stock, par value \$0.0001 per share ("*Class B Common Stock*"), which were issued in a private placement prior to the closing of the Company's initial public offering;

WHEREAS, on December 1, 2020, the Sponsor transferred 25,000 shares of Class B Common Stock held by the Sponsor to each of Kent Schofield, Tripp Jones, Patrick Eggen and Courtney Robinson (collectively, the "*Company Directors*");

WHEREAS, in December 2020, the Company effected a stock dividend of approximately 0.143 of a share for each share of Class B Common Stock outstanding, resulting in an aggregate of 5,750,000 shares of Class B Common Stock outstanding (the outstanding shares of Class B Common Stock held by each of the Sponsor and the Company Directors being referred to herein as the "*Founder Shares*");

WHEREAS, upon the consummation of the transactions (the "*Closing*") contemplated by the Merger Agreement (as defined below), the Founder Shares shall be automatically converted into shares of the Company's Class A common stock, par value \$0.0001 per share ("*Class A Common Stock*"), on a one-for-one basis, subject to adjustment;

WHEREAS, in order to finance the Company's transaction costs in connection with the transactions contemplated by the Merger Agreement, the Sponsor or an affiliate of the Sponsor or certain of the Company's officers and directors may, but are not obligated to, loan to the Company funds as the Company may require, of which up to \$1,500,000 of such loans may be convertible into up to an additional 1,500,000 warrants at a price of \$1.00 per warrant (the "Working Capital Warrants");

WHEREAS, the Company entered into that certain Agreement and Plan of Merger and Reorganization, dated as of December 8, 2022 (the "*Merger Agreement*"), by and among the Company, 7GC Merger Sub I, Inc., a Delaware corporation and an indirect wholly owned subsidiary of the Company ("*First Merger Sub*"), 7GC Merger Sub II, LLC, a Delaware limited liability company and a direct wholly owned subsidiary of the Company ("*Second Merger Sub*" and, together with First Merger Sub, the "*Merger Subs*" and each, a "*Merger Sub*"), and Banzai International, Inc., a Delaware corporation ("*Banzai*"), pursuant to which, through a series of mergers at the Closing with the Merger Subs, Banzai will be merged with and into Second Merger Sub, with Second Merger Sub surviving the mergers and remaining a wholly-owned subsidiary of the Company;

WHEREAS, pursuant to the transactions contemplated by the Merger Agreement and subject to the terms and conditions set forth therein, upon the Closing, the New Holders will receive shares of the Company's Class A Common Stock and/or Class B Common Stock;

WHEREAS, each of the Company and the Existing Holders is a party to that certain Registration Rights Agreement, dated December 22, 2020 (the "*Existing Registration Rights Agreement*"), pursuant to which the Company granted the Existing Holders certain registration rights with respect to certain securities of the Company, as set forth therein; and

WHEREAS, upon the Closing, each party to the Existing Registration Rights Agreement desires to amend and restate the Existing Registration Rights Agreement in its entirety as set forth herein, and the Company and the Holders desire to enter into this Agreement, pursuant to which the Company shall grant the Holders certain registration rights with respect to the Registrable Securities (as defined below) on the terms and conditions set forth in this Agreement.

NOW, **THEREFORE**, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

- 1.1 <u>Definitions</u>. The terms defined in this <u>Article I</u> shall, for all purposes of this Agreement, have the respective meanings set forth below:
- "Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Chief Executive Officer or principal financial officer of the Company, after consultation with counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, declared effective, or used, as the case may be, and (iii) the Company has a bona fide business purpose for not making such information public.
 - "Agreement" shall have the meaning given in the Preamble.
 - "Banzai" shall have the meaning given in the Recitals hereto.
 - "Board" shall mean the Board of Directors of the Company.
- "business day" shall mean a day, other than a Saturday or Sunday, on which commercial banks in San Francisco, California or Salt Lake City, Utah are open for the general transaction of business.
 - "Closing" shall have the meaning given in the Recitals hereto.
 - "Commission" shall mean the U.S. Securities and Exchange Commission.

- "Common Stock" shall mean the common stock, par value \$0.0001 per share, of the Company (which, for the avoidance of doubt, is comprised of shares designated as "Class A" and "Class B" common stock).
 - "Company" shall have the meaning given in the Preamble.
 - "Company Underwritten Demand Notice" shall have the meaning given in subsection 2.1.3.
 - "Demanding Holder" shall have the meaning given in subsection 2.1.3.
 - "Effectiveness Deadline" shall have the meaning given in subsection 2.1.1.
 - "Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.
 - "Existing Holders" shall have the meaning given in the Preamble.
 - "Existing Registration Rights Agreement" shall have the meaning given in the Recitals hereto.
 - "First Merger Sub" shall have the meaning given in the Recitals hereto.
- "Form S-1 Registration Statement" shall mean a Registration Statement on Form S-1 allowing for the delayed or continuous offering and sale of the Registrable Securities pursuant to Rule 415.
- "Form S-3 Shelf" shall mean a Registration Statement on Form S-3 allowing for the delayed or continuous offering and sale of the Registrable Securities pursuant to Rule 415.
- "Founder Shares" shall have the meaning given in the Recitals hereto and shall, for the avoidance of doubt, be deemed to include the shares of Class A Common Stock issuable upon conversion thereof.
- "Founder Shares Lock-Up Period" shall mean, with respect to the Founder Shares, from the date hereof until the earlier of (A) 180 days after the date hereof; and (B) the date on which the Company completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of the Company's stockholders having the right to exchange their shares of Common Stock for cash, securities or other property.
- "GEM Agreement" shall mean that certain Share Purchase Agreement, dated as of May 27, 2022, by and between the Company and GEM Global Yield LLC SCS and GEM Yield Bahamas Limited.
 - "Holder" and "Holders" shall have the meanings given in the Preamble.
- "*Insider Letter*" shall mean that certain letter agreement, dated as of December 22, 2020, by and among the Company, the Sponsor and each of the Company's officers and directors signatory thereto.
 - "Lock-Up Period" shall mean the Founder Shares Lock-Up Period.
 - "Maximum Number of Securities" shall have the meaning given in subsection 2.1.5.
 - "Merger Agreement" shall have the meaning given in the Recitals hereto.

- "Merger Sub" and "Merger Subs" shall have the meanings given in the Recitals hereto.
- "*Misstatement*" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus, or necessary to make the statements in a Registration Statement or Prospectus (in the case of the Prospectus, in the light of the circumstances under which they were made) not misleading.
 - "New Holders" shall have the meaning given in the Preamble.
- "*Permitted Transferees*" shall mean any person or entity (i) to whom a Holder of Registrable Securities is permitted to transfer such Registrable Securities prior to the expiration of the Lock-Up Period, under the Insider Letter, this Agreement and any other applicable agreement between such Holder and the Company, and to any transferee thereafter, and (ii) who agrees to become bound by the transfer restrictions set forth in this Agreement.
 - "Piggyback Registration" shall have the meaning given in subsection 2.2.1.
 - "Pro Rata" shall have the meaning given in subsection 2.1.5.
- "*Prospectus*" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.
- "Registrable Security" shall mean (a) the Founder Shares (including, for the avoidance of doubt, the shares of Class A Common Stock issued or issuable upon the conversion of any Founder Shares), (b) any issued and outstanding share of Common Stock or any other equity security (including the shares of Common Stock issued or issuable upon the exercise of any other equity security) of the Company held by an Existing Holder as of the date of this Agreement, (c) any shares of Common Stock issued or issuable upon exercise of the Working Capital Warrants, (d) any outstanding shares of Common Stock or any other equity security of the Company held by a New Holder as of the date of this Agreement (including shares transferred to a Permitted Transferee and the shares of Common Stock issued or issuable upon the exercise of any such other equity security) and (e) any other equity security of the Company issued or issuable with respect to any such share of the Common Stock described in the foregoing clauses (a) through (e) by way of a stock dividend or stock split or in connection with a combination of shares, distribution, recapitalization, merger, consolidation or reorganization or other similar event; provided, however, that, as to any particular Registrable Securities shall cease to be Registrable Securities upon the earliest to occur of: (i) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (ii) such securities shall have been otherwise transferred, new certificates or book entry positions for such securities not bearing a legend restricting further transfer shall have been otherwise transferred, new certificates or book entry positions for such securities not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not require registr

"*Registration*" shall mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

- "Registration Expenses" shall mean the out-of-pocket expenses of a Registration, including, without limitation, the following:
- (A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Common Stock is then listed;
- (B) fees and expenses of compliance with securities or blue sky laws (including reasonable and customary fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);
 - (C) printing, messenger, telephone and delivery expenses;
 - (D) reasonable fees and disbursements of counsel for the Company;
- (E) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration; and
- (F) reasonable fees and expenses of one (1) legal counsel selected by the majority-in-interest of the Demanding Holders initiating an Underwritten Demand to be registered for offer and sale in the applicable Registration.
- "*Registration Statement*" shall mean any registration statement under the Securities Act that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.
 - "Requesting Holders" shall have the meaning given in subsection 2.1.3.
 - "Restricted Securities" shall have the meaning given in subsection 3.6.1.
- "Rule 144" shall mean Rule 144 promulgated under the Securities Act (together with any successor rule promulgated thereafter by the Commission).
 - "Rule 415" shall have the meaning given in subsection 2.1.1.
 - "Second Merger Sub" shall have the meaning given in the Recitals hereto.
 - "Securities Act" shall mean the Securities Act of 1933, as amended from time to time.
 - "Sponsor" shall have the meaning given in the Preamble.
- "*Underwriter*" shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.
 - "Underwritten Demand" shall have the meaning given in subsection 2.1.3.
 - "Underwritten Demand Notice" shall have the meaning given in subsection 2.1.3.
- "Underwritten Registration" or "Underwritten Offering" shall mean a Registration in which securities of the Company are sold to an Underwritter in a firm commitment underwriting for distribution to the public, including an offering and/or sale of Registrable Securities by any Holder in a block trade or on an underwritten basis (whether firm commitment or otherwise) without substantial marketing efforts prior to pricing, including, without limitation, a same day trade, overnight trade or similar transaction, but excluding a variable price reoffer.
 - "Working Capital Warrants" shall have the meaning given in the Recitals hereto.

ARTICLE II REGISTRATIONS

2.1 Shelf Registration.

2.1.1 Initial Registration. The Company shall use its reasonable best efforts to, as promptly as reasonably practicable, but in no event later than thirty (30) days after the Closing (the "Filing Deadline"), file a Form S-1 Registration Statement under the Securities Act to permit the public resale of all the Registrable Securities held by the Holders (and certain other outstanding equity securities of the Company) from time to time as permitted by Rule 415 under the Securities Act (or any successor or similar provision adopted by the Commission then in effect) ("Rule 415") on the terms and conditions specified in this subsection 2.1.1 and shall use its reasonable best efforts to cause such Registration Statement to be declared effective as promptly as reasonably practicable after the initial filing thereof, but in no event later than sixty (60) business days following the Filing Deadline (the "Effectiveness Deadline"); provided, that the Effectiveness Deadline shall be extended to one hundred and twenty (120) days after the Filing Deadline if the Registration Statement is reviewed by, and receives comments from, the Commission. The Registration Statement filed with the Commission pursuant to this subsection 2.1.1 shall be on a Form S-1 Registration Statement or such other form of registration statement as is then available to effect a registration for resale of such Registrable Securities, covering such Registrable Securities, and shall contain a Prospectus in such form as to permit any Holder to sell such Registrable Securities pursuant to Rule 415 at any time beginning on the effective date for such Registration Statement. A Registration Statement filed pursuant to this subsection 2.1.1 shall provide for the resale pursuant to any method or combination of methods legally available to, and reasonably requested prior to effectiveness by, the Holders. The Company shall use its reasonable best efforts to cause a Registration Statement filed pursuant to this subsection 2.1.1 to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, that another Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities. When effective, a Registration Statement filed pursuant to this subsection 2.1.1 (including the documents incorporated therein by reference) will comply as to form in all material respects with all applicable requirements of the Securities Act and the Exchange Act and will not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading (in the case of any Prospectus contained in such Registration Statement, in the light of the circumstances under which such statement is made).

2.1.2 Form S-3 Shelf. The Company shall use its commercially reasonable efforts to convert the Form S-1 Registration Statement to a Form S-3 Shelf as soon as practicable after the Company is eligible to use a Form S-3 Shelf. If the Company files a Form S-3 Shelf and thereafter the Company becomes ineligible to use Form S-3 for secondary sales, the Company shall use its reasonable best efforts to file a Form S-1 Registration Statement as promptly as reasonably practicable to replace the shelf registration statement that is a Form S-3 Shelf and have the Form S-1 Registration Statement declared effective as promptly as reasonably practicable and to cause such Form S-1 Registration Statement to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, that another Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities.

2.1.3 Underwritten Offering. At any time and from time to time following the effectiveness of the Registration Statement required by subsections 2.1.1 or 2.1.2, any Holder may request to sell all or a portion of their Registrable Securities (a "Demanding Holder") in an underwritten offering that is registered pursuant to such Registration Statement (an "Underwritten Demand"), provided that such Holder(s) (a) reasonably expect aggregate gross proceeds in excess of \$50,000,000 from such Underwritten Offering or (b) reasonably expects to sell all of the Registrable Securities held by such Holder in such Underwritten Offering but in no event less than \$10,000,000 in aggregate gross proceeds. All requests for an Underwritten Offering shall be made by giving written notice to the Company (the "Underwritten Demand Notice"). Each Underwritten Demand Notice shall specify the approximate number of Registrable Securities proposed to be sold in the Underwritten Offering and the expected price range (net of underwriting discounts and commissions) of such Underwritten Offering. Within five (5) business days after receipt of any Underwritten Demand Notice, the Company shall give written notice of such requested Underwritten Offering (the "Company Underwritten Demand Notice") to all other Holders of Registrable Securities (the "Requesting Holders") and, subject to reductions consistent with the Pro Rata calculations in subsection 2.1.5, shall include in such Underwritten Offering all Registrable Securities with respect to which the Company has received written requests for inclusion therein, within five (5) days after sending the Company Underwritten Demand Notice. The Company shall enter into an underwriting agreement in a form as is customary in Underwritten Offerings of securities by the Company with the managing Underwriter or Underwriters selected by the initiating Demanding Holders with the written consent of the Company (such consent not to be unreasonably withheld, conditioned or delayed) and shall take all such other reasonable actions as are requested by the managing Underwriter or Underwriters in order to expedite or facilitate the disposition of such Registrable Securities. In connection with any Underwritten Offering contemplated by this <u>subsection 2.1.3</u>, subject to <u>Section 3.3</u> and <u>Article IV</u>, the underwriting agreement into which each Holder and the Company shall enter shall contain such representations and warranties, covenants, indemnities and other rights and obligations of the Company and such Holders as are customary in underwritten offerings of securities. Under no circumstances shall the Company be obligated to effect (x) more than an aggregate of three (3) Underwritten Offerings pursuant to an Underwritten Demand by the Holders under this subsection 2.1.3 with respect to any or all Registrable Securities held by such Holders and (y) more than two (2) Underwritten Offerings per year pursuant to this subsection 2.1.3; provided, however, that an Underwritten Offering pursuant to an Underwritten Demand shall not be counted for such purposes unless a Registration Statement that may be available at such time has become effective and all of the Registrable Securities requested by the Requesting Holders and the Demanding Holders to be registered on behalf of the Requesting Holders and the Demanding Holders in such Registration Statement have been sold, in accordance with Section 3.1 of this Agreement.

2.1.4 Holder Information Required for Participation in Underwritten Offering. At least ten (10) business days prior to the first anticipated filing date of a Registration Statement pursuant to this Article II, the Company shall use reasonable best efforts to notify each Holder in writing (which may be by email) of the information reasonably necessary about the Holder to include such Holder's Registrable Securities in such Registration Statement. Notwithstanding anything else in this Agreement, the Company shall not be obligated to include such Holder's Registrable Securities to the extent the Company has not received such information, and received any other reasonably requested agreements or certificates, on or prior to the fifth business day prior to the first anticipated filing date of a Registration Statement pursuant to this Article II.

2.1.5 Reduction of Underwritten Offering. If the managing Underwriter or Underwriters in an Underwritten Offering, in good faith, advises the Company, the Demanding Holders and the Requesting Holders (if any) in writing that the dollar amount or number of Registrable Securities that the Demanding Holders and the Requesting Holders (if any) desire to sell, taken together with all other Common Stock or other equity securities that the Company desires to sell and the Common Stock, if any, as to which a Registration has been requested pursuant to separate written contractual piggyback registration rights held by any other stockholders who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders (pro rata based on the respective number of Registrable Securities that each Demanding Holder has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders have requested be included in such Underwritten Registration (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Registrable Securities of Requesting Holders (Pro Rata, based on the respective number of Registrable Securities that each Requesting Holder has so requested) exercising their rights to register their Registrable Securities pursuant to subsection 2.1.3 hereof, without exceeding the Maximum Number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), (ii) and (iii), the Common Stock or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities.

2.1.6 <u>Underwritten Offering Withdrawal</u>. A majority-in-interest of the Demanding Holders initiating an Underwritten Demand or a majority-in-interest of the Requesting Holders (if any), pursuant to a Registration under <u>subsection 2.1.3</u> shall have the right to withdraw from a Registration pursuant to an Underwritten Offering pursuant to <u>subsection 2.1.3</u> for any or no reason whatsoever upon written notification to the Company and the Underwriters (if any) of their intention to withdraw from such Registration at least five (5) business days prior to the effectiveness of the Registration Statement filed with the Commission with respect to the Registration of their Registrable Securities pursuant to such Underwritten Offering (or in the case of an Underwritten Registration pursuant to Rule 415, at least five (5) business days prior to the time of pricing of the applicable offering). Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to an Underwritten Offering prior to its withdrawal under this <u>subsection 2.1.6</u>.

2.2 Piggyback Registration.

2.2.1 Piggyback Rights. If, at any time on or after the Closing, the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company, including, without limitation, pursuant to Section 2.1 hereof), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for a rights offering or an exchange offer or offering of securities solely to the Company's existing stockholders, (iii) for an offering of debt that is convertible into equity securities of the Company or (iv) for a dividend reinvestment plan, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than ten (10) days before the anticipated filing date of such Registration Statement, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (such Registration a "Piggyback Registration"). The Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and shall use its reasonable best efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering to permit the Registrable Securities requested by the Holders pursuant to this subsection 2.2.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company included in such Registration and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this subsection 2.2.1 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.

- 2.2.2 <u>Reduction of Piggyback Registration</u>. If the managing Underwriter or Underwriters in an Underwritten Registration that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of the shares of Common Stock that the Company desires to sell, taken together with (i) the shares of Common Stock, if any, as to which Registration has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder, (ii) the Registrable Securities as to which registration has been requested pursuant to <u>Section 2.2</u> hereof, and (iii) the shares of Common Stock, if any, as to which Registration has been requested pursuant to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:
- (a) If the Registration is undertaken for the Company's account, the Company shall include in any such Registration (A) first, the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to <u>subsection 2.2.1</u> hereof, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the Common Stock, if any, as to which Registration has been requested or demanded pursuant to written contractual Piggyback Registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities;
- (b) If the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration (A) first, the Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A), (B) and (C), the Common Stock or other equity securities for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities.
- 2.2.3 <u>Piggyback Registration Withdrawal</u>. Any Holder of Registrable Securities shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration at least five (5) business days prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Piggyback Registration (or in the case of an Underwritten Registration pursuant to Rule 415, at least five (5) business days prior to the time of pricing of the applicable offering). The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this <u>subsection 2.2.3</u>.
- 2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to <u>Section 2.2</u> hereof shall not be counted as a Registration pursuant to an Underwritten Offering effected under <u>subsection 2.1.3</u>.
- 2.3 Restrictions on Registration Rights. Notwithstanding anything to the contrary contained herein, the Company shall not be obligated to (but may, at its sole option) (A) effect an Underwritten Offering (i) within sixty (60) days after the closing of an Underwritten Offering, or (ii) during the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, a Company-initiated Registration and provided that the Company has delivered written notice to the Holders prior to receipt of an Underwritten Demand pursuant to subsection 2.1.3 and it continues to actively employ, in good faith, all reasonable best efforts to cause the applicable Registration Statement to become effective or (B) file a Registration Statement (or any amendment thereto) or effect an Underwritten Offering (or, if the Company has filed a shelf Registration Statement and has included Registrable Securities therein, the Company shall be entitled to suspend the offer and sale of Registrable Securities pursuant to such Registration Statement) for a period of up to forty-five (45) days (i) if the Holders have requested an Underwritten Demand and the Company and the Holders are unable to obtain the commitment of Underwriters to firmly underwrite the offer; or (ii) in the good faith judgment of the Board such Underwritten Offering would be materially detrimental to the Company and the Board concludes as a result that it is essential to defer the filing of such Registration Statement at such time, provided that in each case of (i) and (ii) the Company shall furnish to such Holders a certificate signed by the Chairman of the Board stating that in the good faith judgment of the Board it would be materially detrimental to the Company for such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement and provided, further, that the Compa

ARTICLE III COMPANY PROCEDURES

- 3.1 <u>General Procedures</u>. If the Company is required to effect the Registration of Registrable Securities, subject to applicable law and any regulations promulgated by any securities exchange on which the Registrable Securities are then listed, the Company shall use its reasonable best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall, as expeditiously as possible:
- 3.1.1 prepare and file with the Commission within thirty (30) days a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold;
- 3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be reasonably requested by any Holder or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus;
- 3.1.3 prior to filing a Registration Statement or the Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and each Holder of Registrable Securities included in such Registration, and such Holder's legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and each Holder of Registrable Securities included in such Registration or the legal counsel for any such Holders may reasonably request in order to facilitate the disposition of the Registrable Securities owned by such Holders;
- 3.1.4 prior to any public offering of Registrable Securities, use its reasonable best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as any Holder of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business or as a dealer in securities in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;
- 3.1.5 cause all such Registrable Securities to be listed on each securities exchange or automated quotation system on which similar securities issued by the Company are then listed;
- 3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;
 - 3.1.7 provide a CUSIP number for all Registrable Securities, not later than the effective date of such Registration Statement;
- 3.1.8 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;
- 3.1.9 at least five (5) days prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus or any document that is to be incorporated by reference into such Registration Statement or Prospectus, furnish a copy thereof to each seller of such Registrable Securities and its counsel, including, without limitation, providing copies promptly upon receipt of any comment letters received with respect to any such Registration Statement or Prospectus;

- 3.1.10 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in Section 3.4 hereof;
- 3.1.11 permit a representative of the Holders (such representative to be selected by a majority of the participating Holders), the Underwriter(s), if any, and any attorney or accountant retained by such Holders or Underwriter(s) to participate, at each such person's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; provided, however, that any such representative or Underwriter enters into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information; and provided further, the Company may not include the name of any Holder or Underwriter or any information regarding any Holder or Underwriter in any Registration Statement or Prospectus, any amendment or supplement to such Registration Statement or Prospectus, any document that is to be incorporated by reference into such Registration Statement or Prospectus, or any response to any comment letter, without the prior written consent of such Holder or Underwriter and providing each such Holder or Underwriter a reasonable amount of time to review and comment on such applicable document, which comments the Company shall include unless contrary to applicable law;
- 3.1.12 obtain a "comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Registration which the participating Holders may rely on, in customary form and covering such matters of the type customarily covered by "comfort" letters as the managing Underwriter may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders;
- 3.1.13 on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion and negative assurance letter, dated as of such date, of counsel representing the Company for the purposes of such Registration, addressed to the Holders, the placement agent or sales agent, if any, and the Underwriter(s), if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Holders, placement agent, sales agent, or Underwriter(s) may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to a majority in interest of the participating Holders;
- 3.1.14 in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;
- 3.1.15 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission), which requirement will be deemed satisfied if the Company timely files Forms 10-Q and 10-K, as may be required to be filed under the Exchange Act and otherwise complies with Rule 158 under the Securities Act;

- 3.1.16 if the Registration involves the Registration of Registrable Securities involving gross proceeds in excess of \$50,000,000, use its reasonable best efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter(s) in any Underwritten Offering; and
- 3.1.17 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders, in connection with such Registration.
- 3.2 <u>Registration Expenses</u>. Except as otherwise provided herein, the Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "Registration Expenses," all reasonable fees and expenses of any legal counsel representing any Holder.
- 3.3 Requirements for Participation in Underwritten Offerings. No person or entity may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person or entity (i) agrees to sell such person's or entity's securities on the basis provided in any underwriting arrangements approved by the Company and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements. Notwithstanding anything in this Agreement to the contrary, if any Holder does not provide to the Company in writing information and affidavits as the Company reasonably requests for use in connection with any Registration Statement or Prospectus, the Company may exclude such Holder's Registrable Securities from the applicable Registration Statement or Prospectus if the Company determines, based on the advice of outside legal counsel, that such information is necessary to effect the registration and such Holder continues thereafter to withhold such information. For the avoidance of doubt, the exclusion of a Holder's Registrable Securities as a result of this Section 3.3 shall not affect the registration of the other Registrable Securities to be included in such Registration.
- 3.4 <u>Suspension of Sales</u>; <u>Adverse Disclosure</u>. Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until it has received copies of a supplemented or amended Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice), or until it is advised in writing by the Company that the use of the Prospectus may be resumed. If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would require the Company to make an Adverse Disclosure or would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time, but in no event more than forty-five (45) days, determined in good faith by the Company to be necessary for such purpose. In the event the Company exercises its rights under the preceding sentence, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities. The Company shall immediately notify the Holders of the expiration of any period during which it exercised its rights under this Section 3.4.

3.5 Reporting Obligations. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell shares of the Common Stock held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144, including providing any legal opinions. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

3.6 Lock-Up Restrictions.

3.6.1 During the Lock-Up Period, none of the Existing Holders shall offer, sell, contract to sell, pledge, grant any option to purchase, make any short sale or otherwise dispose of or distribute any shares of Common Stock that are subject to the Lock-Up Period or any securities convertible into, exercisable for, exchangeable for or that represent the right to receive shares of Common Stock that are subject to the Lock-Up Period, whether now owned or hereinafter acquired, that are owned directly by such Existing Holder (including securities held as a custodian) or with respect to which such Existing Holder has beneficial ownership within the rules and regulations of the Commission (such securities that are subject to the Lock-Up Period, the "Restricted Securities"), other than any transfer to an affiliate of an Existing Holder or to a Permitted Transferee, as applicable. The foregoing restriction is expressly agreed to preclude each Existing Holder, as applicable, from engaging in any hedging or other transaction with respect to Restricted Securities which is designed to or which reasonably could be expected to lead to or result in a sale or disposition of the Restricted Securities even if such Restricted Securities would be disposed of by someone other than such Existing Holder. Such prohibited hedging or other transactions include any short sale or any purchase, sale or grant of any right (including any put or call option) with respect to any of the Restricted Securities of the applicable Existing Holder, or with respect to any security that includes, relates to, or derives any significant part of its value from such Restricted Securities.

3.6.2 Each Existing Holder hereby represents and warrants that it now has and, except as contemplated by this <u>subsection 3.6.2</u> for the duration of the Lock-Up Period, will have good and marketable title to its Restricted Securities, free and clear of all liens, encumbrances, and claims that could impact the ability of such Existing Holder to comply with the foregoing restrictions Each Existing Holder agrees and consents to the entry of stop transfer instructions with the Company's transfer agent and registrar against the transfer of any Restricted Securities during the Lock-Up Period.

ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

4.1 Indemnification.

4.1.1 The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers and directors and each person who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and out-of-pocket expenses (including reasonable and documented attorneys' fees) caused by any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, except insofar as the same are caused by or contained in any information furnished in writing to the Company by such Holder expressly for use therein. The Company shall indemnify the Underwriters, their officers and directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.

- 4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by law, shall indemnify the Company, its directors and officers and agents and each person who controls the Company (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and out-of-pocket expenses (including without limitation reasonable attorneys' fees) resulting from any untrue statement of material fact contained in the Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information or affidavit so furnished in writing by such Holder expressly for use therein; provided, however, that the obligation to indemnify under this Section 4.1.2 shall be several, not joint and several, among the Holders of Registrable Securities, and the total liability of a Holder under this Section 4.1.2 shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement. The Holders of Registrable Securities shall indemnify the Underwriters, their officers, directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to indemnification of the Company.
- 4.1.3 Any person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided, that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party), and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel (plus local counsel) for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnified party, or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.
- 4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling person of such indemnified party and shall survive the transfer of securities.
- 4.1.5 If the indemnification provided under Section 4.1 hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; provided, however, that the liability of any Holder under this subsection 4.1.5 shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in subsections 4.1.1, 4.1.2 and 4.1.3 above, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this subsection 4.1.5 were determined by Pro Rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this subsection 4.1.5. from any person who was not guilt

ARTICLE V MISCELLANEOUS

5.1 Notices. Any notice or communication under this Agreement must be in writing and given by (i) deposit in the United States mail, addressed to the party to be notified, postage prepaid and registered or certified with return receipt requested, (ii) delivery in person or by courier service providing evidence of delivery, or (iii) transmission by hand delivery, facsimile or electronic mail. Each notice or communication that is mailed, delivered, or transmitted in the manner described above shall be deemed sufficiently given, served, sent, and received, in the case of mailed notices, on the third business day following the date on which it is mailed and, in the case of notices delivered by courier service, hand delivery, facsimile or electronic mail, at such time as it is delivered to the addressee (except in the case of electronic mail, with the delivery receipt or the affidavit of messenger) or at such time as delivery is refused by the addressee upon presentation. Any notice or communication under this Agreement must be addressed, if to the Company, to: [Banzai Holdings, Inc.], 435 Ericksen Ave, Suite 250, Bainbridge Island, WA 98110, Attn: Joseph Davy, E-mail: joe@banzai.io, with a copy to: Cooley LLP, 1700 Seventh Avenue, Suite 1900, Seattle, WA 98101-1355, Attn: Sonya Erickson, E-mail: serickson@cooley.com, and, if to any Holder, at such Holder's address or contact information as set forth in the Company's books and records. Any party may change its address for notice at any time and from time to time by written notice to the other parties hereto, and such change of address shall become effective thirty (30) days after delivery of such notice as provided in this Section 5.1.

5.2 Assignment; No Third Party Beneficiaries.

5.2.1 This Agreement and the rights, duties and obligations of the Company and the Holder of Registrable Securities, as the case may be, hereunder may not be assigned or delegated by the Company or the Holders of Registrable Securities, as the case may be, in whole or in part, except in connection with a transfer of Registrable Securities by such Holder to a Permitted Transferee but only if such Permitted Transferee agrees to become bound by the transfer restrictions set forth in this Agreement.

5.2.2 Prior to the expiration of the Lock-Up Period, no Holder subject to the Lock-Up Period may assign or delegate such Holder's rights, duties or obligations under this Agreement, in whole or in part, in violation of the Lock-Up Period, except in connection with a transfer of Registrable Securities by such Holder to a Permitted Transferee but only if such Permitted Transferee agrees to become bound by the transfer restrictions set forth in this Agreement.

- 5.2.3 This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors and the permitted assigns of the Holders, which shall include Permitted Transferees.
- 5.2.4 This Agreement shall not confer any rights or benefits on any persons that are not parties hereto, other than as expressly set forth in this Agreement and Section 5.2 hereof.
- 5.2.5 No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in <u>Section 5.1</u> hereof and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any transfer or assignment made other than as provided in this <u>Section 5.2</u> shall be null and void.
- 5.3 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts (including facsimile or PDF counterparts), each of which shall be deemed an original, and all of which together shall constitute the same instrument, but only one of which need be produced. Each party agrees that an electronic copy of this Agreement shall be considered and treated like an original, and that an electronic or digital signature shall be as valid as a handwritten signature (including .pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000 (e.g., www.docusign.com)).
- 5.4 <u>Governing Law; Venue.</u> NOTWITHSTANDING THE PLACE WHERE THIS AGREEMENT MAY BE EXECUTED BY ANY OF THE PARTIES HERETO, THE PARTIES EXPRESSLY AGREE THAT (I) THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF NEW YORK AS APPLIED TO AGREEMENTS AMONG NEW YORK RESIDENTS ENTERED INTO AND TO BE PERFORMED ENTIRELY WITHIN NEW YORK, WITHOUT REGARD TO THE CONFLICT OF LAW PROVISIONS OF SUCH JURISDICTION AND (II) THE VENUE FOR ANY ACTION TAKEN WITH RESPECT TO THIS AGREEMENT SHALL BE ANY STATE OR FEDERAL COURT IN NEW YORK COUNTY IN THE STATE OF NEW YORK.
- 5.5 Amendments and Modifications. Upon the written consent of the Company and the Holders of at least a majority in interest of the Registrable Securities at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; provided, however, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects either the Existing Holders as a group or the New Holders as a group, as the case may be, in a manner that is materially adversely different from the New Holders or the Existing Holders, respectively, shall require the consent of at least a majority-in-interest of the Registrable Securities held by such Existing Holders or a majority-in-interest of the Registrable Securities held by the New Holders, as applicable, at the time in question so affected; provided, further, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects one Holder or group of affiliated Holders, solely in its capacity as a holder of the shares of capital stock of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder or group of affiliated Holders so affected. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party. Any amendment, termination, or waiver effected in accordance with this Section 5.5 shall be binding on each party hereto and all of such party's successors and permitted assigns, regardless of whether or no

5.6 Other Registration Rights. The Company represents and warrants that no person, other than (i) a Holder of Registrable Securities and (ii) a holder of securities of the Company that are registrable pursuant to the GEM Agreement, has any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities for its own account or for the account of any other person. Further, the Company represents and warrants that, except with respect to the GEM Agreement, this Agreement supersedes the Existing Registration Rights Agreement and any other registration rights agreement or agreement with similar terms and conditions and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail. Notwithstanding the foregoing, the Company and the Holders hereby acknowledge that the Company has granted resale registration rights to certain holders of Company securities in the GEM Agreement, and that nothing herein shall restrict the ability of the Company to fulfill its resale registration obligations under the GEM Agreement.

5.7 Opt-Out Requests. Each Holder shall have the right, at any time and from time to time (including after receiving information regarding any potential public offering), to elect to not receive any notice that the Company or any other Holders otherwise are required to deliver pursuant to this Agreement by delivering to the Company a written statement signed by such Holder that it does not want to receive any notices hereunder (an "Opt-Out Request"); in which case and notwithstanding anything to the contrary in this Agreement the Company and other Holders shall not be required to, and shall not, deliver any notice or other information required to be provided to Holders hereunder to the extent that the Company or such other Holders reasonably expect would result in a Holder acquiring material non-public information within the meaning of Regulation FD promulgated under the Exchange Act. An Opt-Out Request may state a date on which it expires or, if no such date is specified, shall remain in effect indefinitely. A Holder who previously has given the Company an Opt-Out Request may revoke such request at any time, and there shall be no limit on the ability of a Holder to issue and revoke subsequent Opt-Out Requests; provided that each Holder shall use commercially reasonable efforts to minimize the administrative burden on the Company arising in connection with any such Opt-Out Requests.

5.8 Existing Registration Rights Agreement. The Sponsor and the Existing Holders hereby agree that upon execution of this Agreement by the Sponsor and the Existing Holders, the Existing Registration Rights Agreement shall be automatically terminated and superseded in its entirety by this Agreement.

5.9 Term. This Agreement shall terminate upon the earlier of (i) the seventh (7th) anniversary of the date of this Agreement, and (ii) the date as of which all of the Registrable Securities have been sold or disposed of or, (iii) with respect to any particular holder, (A) such Holder is permitted to sell the Registrable Securities held by him, her or it under Rule 144 (or any similar provision) under the Securities Act without limitation on the amount of securities sold or the manner of sale and without compliance with the current public reporting requirements set forth under Rule 144(i)(2), or (B) such Holder requests and the Company agrees to authorize the Company's transfer agent to remove any legend on share certificates of such Holder's Registrable Securities restricting further transfer (or any similar restriction in book entry positions of such Holder). The provisions of Section 3.5 and Article IV shall survive any termination.

5.8 Entire Agreement. This Agreement constitutes the entire understanding and agreement between the parties as to the matters covered herein and supersedes and replaces any prior understanding, agreement or statement of intent, in each case, written or oral, of any and every nature with respect thereto.

[Signature Page Follows]

	COMPANY:
	[BANZAI HOLDINGS INC.], a Delaware corporation
	By:
	Name: Title:
	SPONSOR:
	7GC & CO. HOLDINGS LLC, a Delaware limited liability company
	By:
	Name: Title:
	OTHER PRE-IPO HOLDERS:
	Courtney Robinson
	Tripp Jones
	Kent Schofield
	Patrick Eggen
	NEW HOLDERS:
	[●]
[Signature Page	e to Amended and Restated Registration Rights Agreement]

SPONSOR FORFEITURE AGREEMENT

August 4, 2023

7GC & Co. Holdings Inc. c/o 7GC & Co. Holdings LLC 388 Market Street, Suite 1300 San Francisco, CA 94111 Attention: Jack Leeney and Chris Walsh

E-mail: jack@7GC.co and chris@7GC.co

Banzai International, Inc. 435 Ericksen Ave., Suite 250 Bainbridge Island, WA 98110 Attention: Joseph Davy E-mail: joe@banzai.io

Re: Forfeiture of Certain Sponsor Private Placement Warrants

Ladies and Gentlemen:

Reference is hereby made to: (i) that certain Agreement and Plan of Merger and Reorganization (the "Original Merger Agreement"), dated as of December 8, 2022, by and among Banzai International, Inc., a Delaware corporation (the "Company"), 7GC & Co. Holdings Inc., a Delaware corporation ("7GC"), 7GC Merger Sub I, Inc., a Delaware corporation and an indirect wholly owned subsidiary of 7GC, and 7GC Merger Sub II, LLC, a Delaware limited liability company and a direct wholly owned subsidiary of 7GC, as amended by that certain Amendment to the Agreement and Plan of Merger and Reorganization, dated as of the date hereof (the "Merger Agreement Amendment" and the Original Merger Agreement, as amended by the Merger Agreement, the "Merger Agreement"), by and between 7GC and the Company; and (ii) that certain Private Placement Warrants Purchase Agreement, dated as of December 22, 2020 (as it may from time to time be amended and including all exhibits referenced herein, the "Warrant Purchase Agreement"), by and between 7GC and 7GC & Co. Holdings LLC, a Delaware limited liability company (the "Sponsor"), pursuant to which the Sponsor acquired from 7GC 7,350,000 warrants to purchase shares of 7GC's Class A common stock (the "Sponsor Warrants"), as more specifically set forth therein. Capitalized terms used but not defined herein shall have the meanings ascribed to them in the Merger Agreement or the Warrant Purchase Agreement, as applicable.

In order to induce 7GC and the Company to enter into the Merger Agreement Amendment and to proceed with the transactions contemplated in the Merger Agreement and thereby, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, 7GC, the Company and the Sponsor, hereby agree, pursuant to this letter agreement (this "<u>Letter Agreement</u>"), as follows:

1. Effective immediately prior to (and contingent upon the occurrence) the Closing, the Sponsor Warrants will be automatically forfeited and be of no further force or effect (the "Warrant Forfeiture").

- 2. To effect the Warrant Forfeiture, immediately prior to (and contingent upon) the Closing:
 - a. the Sponsor shall surrender the Sponsor Warrants to 7GC for cancellation and in exchange for no consideration;
- b. 7GC shall immediately retire and cancel all of the Sponsor Warrants (and shall direct 7GC's transfer agent (or such other intermediaries as appropriate) to take any and all such actions incident thereto); and
- c. the Sponsor and 7GC each shall take such actions as are necessary to cause the Sponsor Warrants to be retired and cancelled, after which the Sponsor Warrants shall no longer be issued, outstanding, convertible, or exercisable, and the Sponsor shall provide the Company with evidence that such retirement and cancellation has occurred.
- 3. The Sponsor hereby represents and warrants to 7GC and the Company, as of the date hereof and as of immediately prior to the Warrant Forfeiture, that the Sponsor owns, and holds of record, all of the Sponsor Warrants, free and clear of all Liens and other obligations in respect of the Sponsor Warrants.
- 4. No party hereto may assign either this Letter Agreement or any of its rights, interests, or obligations hereunder without the prior written consent of each of the other parties hereto. Any purported assignment in violation of this paragraph shall be void and ineffectual and shall not operate to transfer or assign any interest or title to the purported assignee. This Letter Agreement shall be binding on the Sponsor, 7GC and the Company, and their respective successors and assigns.
- 5. Any notice, consent, or request to be given in connection with any of the terms or provisions of this Letter Agreement shall be in writing and shall be sent by express mail or similar private courier service, by certified mail (return receipt requested), by hand delivery or facsimile transmission.
- 6. This Letter Agreement shall immediately terminate, without any further action by the parties hereto, at such time, if at all, that the Merger Agreement is terminated in accordance with its terms.
- 7. Section 8.5 (Governing Law), Section 8.10 (Severability), Section 8.11 (Counterparts; Electronic Signatures), Section 8.15 (Waiver of Jury Trial) and Section 8.16 (Jurisdiction) of the Merger Agreement are hereby incorporated into this Letter Agreement, *mutatis mutandis*, as though set out in their entirety in this paragraph 7.

[Signature pages to follow]

In Witness Whereof, this Letter Agreement has been duly executed and delivered by each Party as of the date first above written.

7GC & CO. HOLDINGS INC.

/s/ Jack Leeney By:

Name: Jack Leeney

Title: Chairman and Chief Executive Officer

7GC & CO. HOLDINGS LLC

VII Co-Invest Sponsor LLC, as the

managing member of 7GC & Co. Holdings

LLC

SP Global Advisors LLC, as a manager By:

of VII Co-Invest Sponsor LLC

/s/ Jack Leeney By: Name: Jack Leeney

Title: Manager

BANZAI INTERNATIONAL, INC.

/s/ Joseph P. Davy Name: Joseph P. Davy

Title: Chief Executive Officer

[Signature Page to Sponsor Forfeiture Agreement]





Banzai and 7GC & Co. Holdings Inc. Announce Amendment to Business Combination Agreement

SEATTLE, WA and SAN FRANCISCO, CA – August 7, 2023 – Banzai International, Inc. ("Banzai" or the "Company"), a leading marketing technology company that provides data-driven marketing and sales solutions, and 7GC & Co. Holdings Inc. (**NASDAQ: VII**) ("VII" or "7GC"), a publicly-traded special purpose acquisition company, announced today that they have entered into an amendment (the "Amendment") to their previously disclosed definitive business combination agreement dated December 8, 2022 (the "Business Combination").

The Amendment provides, among other things, that (i) the closing of Banzai's acquisition of Hyros Inc. ("Hyros") is no longer a condition to the closing of the Business Combination, (ii) the total consideration payable to Banzai stockholders is changed to \$100 million based on the standalone value of Banzai, with no right to an earn-out as was contemplated under the original Business Combination agreement, (iii) the "outside date" for the closing of the Business Combination is extended to December 28, 2023, and (iv) the minimum aggregate transaction proceeds condition is replaced with a minimum net cash closing condition, pursuant to which the net cash of 7GC, Banzai and their subsidiaries, after redemptions by 7GC public stockholders and payment of both parties' transaction expenses, must equal or exceed \$5 million.

Banzai had previously agreed to acquire Hyros in a merger that was intended to close immediately prior to the closing of the Business Combination. Banzai and Hyros agreed to terminate the Hyros Merger Agreement on August 1, 2023. The Company remains committed to its inorganic growth strategy once publicly traded.

In connection with the Amendment, 7GC & Co. Holdings LLC (the "Sponsor") agreed, contingent upon closing of the of the Business Combination, to forfeit all 7,350,000 of its private placement warrants acquired in connection with the 7GC initial public offering. At the closing of the Business Combination, such forfeited private placement warrants will be transferred by the Sponsor to 7GC and cancelled in exchange for no consideration.

About 7GC & Co. Holdings

7GC & Co Holdings is a special purpose acquisition company traded on the Nasdaq under the ticker: VII. The firm is a partnership between 7GC, a technology growth fund based in San Francisco, California and Berlin, Germany and Hennessy Capital, and a leading independent SPAC sponsor based in Wilson, Wyoming and Los Angeles, California. 7GC is led by its Chief Executive Officer, Jack Leeney, and its Chief Financial Officer, Christopher Walsh.

About Banzai

Banzai is a marketing technology company that provides data-driven marketing and sales solutions for businesses of all sizes. On a mission to help their customers accomplish their mission, Banzai enables companies of all sizes to target, engage, and measure both new and existing customers more effectively. Banzai customers include Square, Hewlett Packard Enterprise, Thermo Fisher Scientific, Thinkific, Doodle and ActiveCampaign, among thousands of others. Learn more at www.banzai.io.





Important Information for Investors and Stockholders

The Business Combination will be submitted to stockholders of VII for their consideration and approval at a special meeting of stockholders. VII and Banzai will prepare a registration statement on Form S-4 (the "Registration Statement") to be filed with the Securities and Exchange Commission (the "SEC") by VII, which will include preliminary and definitive proxy statements to be distributed to VII's stockholders in connection with VII's solicitation for proxies for the vote by VII's stockholders in connection with the Business Combination and other matters as described in the Registration Statement, as well as the prospectus relating to the offer of the securities to be issued to VII's stockholders and certain of Banzai's equityholders in connection with the completion of the Business Combination. After the Registration Statement has been filed and declared effective, VII will mail a definitive proxy statement and other relevant documents to its stockholders as of the record date established for voting on the Business Combination. VII's stockholders and other interested persons are advised to read, once available, the preliminary proxy statement/prospectus and any amendments thereto and, once available, the definitive proxy statement/prospectus, in connection with VII's solicitation of proxies for its special meeting of stockholders to be held to approve, among other things, the Business Combination, because these documents will contain important information about VII, Banzai and the Business Combination. Stockholders may also obtain a copy of the preliminary or definitive proxy statement, once available, as well as other documents filed with the SEC regarding the Business Combination and other documents filed with the SEC by VII, without charge, at the SEC's website located at www.sec.gov. Copies of these filings may be obtained free of charge on VII's "SEC Filings" website at www.7qc.holdings/sec-filings or by directing a request to info@7qc.co.

No Offer or Solicitation

This press release shall not constitute an offer to sell, or a solicitation of an offer to buy, or a recommendation to purchase, any securities in any jurisdiction, or the solicitation of any vote, consent or approval in any jurisdiction in connection with the Business Combination or any related transactions, nor shall there be any sale, issuance or transfer of any securities in any jurisdiction where, or to any person to whom, such offer, solicitation or sale may be unlawful under the laws of such jurisdiction. This press release does not constitute either advice or a recommendation regarding any securities. No offering of securities shall be made except by means of a prospectus meeting the requirements of the Securities Act of 1933, as amended, or an exemption therefrom.

Participants in the Solicitation

VII and Banzai and their respective directors and executive officers, under SEC rules, may be deemed to be participants in the solicitation of proxies of VII's stockholders in connection with the Business Combination. Investors and security holders may obtain more detailed information regarding VII's directors and executive officers in VII's filings with the SEC, including VII's Annual Report on Form 10-K filed with the SEC on March 31, 2023. Information regarding the persons who may, under SEC rules, be deemed participants in the solicitation of proxies to VII's stockholders in connection with the Business Combination, including a description of their direct and indirect interests, which may, in some cases, be different than those of VII's stockholders generally, will be set forth in the Registration Statement. Stockholders, potential investors and other interested persons should read the Registration Statement carefully when it becomes available before making any voting or investment decisions.

This press release is not a substitute for the Registration Statement or for any other document that VII may file with the SEC in connection with the potential Business Combination. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THE DOCUMENTS FILED WITH THE SEC CAREFULLY AND IN THEIR ENTIRETY WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain free copies of other documents filed with the SEC by VII through the website maintained by the SEC at www.sec.gov.

℃ & Co. Holdings



Forward-Looking Statements

Certain statements included in this press release are not historical facts but are forward-looking statements, including for purposes of the safe harbor provisions under the United States Private Securities Litigation Reform Act of 1995. Forward-looking statements generally are accompanied by words such as "believe," "may," "will," "estimate," "continue," "anticipate," "intend," "expect," "should," "would," "project," "forecast," "predict," "potential," "seem," "seek," "future," "outlook," "target," and similar expressions that predict or indicate future events or trends or that are not statements of historical matters, but the absence of these words does not mean that a statement is not forward-looking. These forward-looking statements include, but are not limited to, (1) references with respect to the anticipated benefits of the Business Combination and anticipated closing timing; (2) the sources and uses of funds for the Business Combination, (3) the anticipated capitalization and enterprise value of the combined company following the consummation of the Business Combination; and (4) current and future potential commercial and customer relationships. These statements are based on various assumptions, whether or not identified in this press release, and on the current expectations of VII's and Banzai's management and are not predictions of actual performance. These forward-looking statements are provided for illustrative purposes only and are not intended to serve as, and must not be relied on by any investor as, a guarantee, an assurance, a prediction or a definitive statement of fact or probability. Actual events and circumstances are difficult or impossible to predict and will differ from assumptions. Many actual events and circumstances are beyond the control of VII and Banzai. These forwardlooking statements are subject to a number of risks and uncertainties, including: the inability of the parties to timely or successfully complete the Business Combination, including due to redemptions by VII's public stockholders, the failure to satisfy the minimum net cash closing condition, the failure to maintain the Nasdaq listing of VII's securities, the failure to obtain certain regulatory approvals or the satisfaction of other conditions to closing in the Merger Agreement; changes in domestic and foreign business, market, financial, political and legal conditions; the inability of the parties to successfully or timely consummate the Business Combination, including the risk that any required stockholder or regulatory approvals are not obtained, are delayed or are subject to unanticipated conditions that could adversely affect the combined company; failure to realize the anticipated benefits of the Business Combination; risks relating to the uncertainty of the projected financial information with respect to Banzai; Banzai's ability to successfully and timely develop, sell and expand its technology and products, and otherwise implement its growth strategy; risks relating to Banzai's operations and business, including information technology and cybersecurity risks, loss of customers and deterioration in relationships between Banzai and its employees; risks related to increased competition; risks relating to potential disruption of current plans, operations and infrastructure of Banzai as a result of the announcement and consummation of the Business Combination; risks that the post-combination company experiences difficulties managing its growth and expanding operations; the amount of redemption requests made by VII's stockholders; the impact of geopolitical, macroeconomic and market conditions, including the COVID-19 pandemic; the ability to successfully select, execute or integrate future acquisitions into the business, which could result in material adverse effects to operations and financial conditions; and those factors discussed in the sections entitled "Risk Factors" and "Special Note Regarding Forward-Looking Statements" in VII's Quarterly Report on Form 10-Q for the quarter ended March 31, 2023, VII's Annual Report on Form 10-K for the year ended December 31, 2022, and in those documents that VII has filed, or will file, with the SEC. If any of these risks materialize or our assumptions prove incorrect, actual results could differ materially from the results implied by these forward-looking statements. The risks and uncertainties above are not exhaustive, and there may be additional risks that neither VII nor Banzai presently know or that VII and Banzai currently believe are immaterial that could also cause actual results to differ from those contained in the forward-looking statements. In addition, forward-looking statements reflect VII's and Banzai's expectations, plans or forecasts of future events and views as of the date of this press release. VII and Banzai anticipate that subsequent events and developments will cause VII's and Banzai's assessments to change. However, while VII and Banzai may elect to update these forward-looking statements at some point in the future, VII and Banzai specifically disclaim any obligation to do so. These forward-looking statements should not be relied upon as representing VII's and Banzai's assessments as of any date subsequent to the date of this press release. Accordingly, undue reliance should not be placed upon the forward-looking statements.

Contacts:

Investors

Cody Slach, Ralf Esper Gateway Group 949-574-3860 VII@gateway-grp.com

Media

Robert Collins Gateway Group 617-797-1979 VII@gateway-grp.com